\begin{table}%t5
\caption{\label{tab:YSO_parameters_L1641} Parameters for YSOs in L1641. Column 4: $^a$ the values estimated from our VIMOS spectra; $^b$ the values from \citet{1995PhDT..........A}; $^c$ the values from \citet{2008AetA...489.1409G}. Column 6: stellar bolometric luminosity derived from optical SED fitting (see Sect.~\ref{sec:analysis:optical_sed_fitting}). Column 9: classification based on the criteria in Appendix~\ref{boundary_sec}. Column 11: aggregate/cluster population or distributed population. Column 13-14: stellar masses and ages estimated using PMS evolutionary tracks from \citet{2008ApJS..178...89D}. Column 16-18: accretion rates estimated with  H$\alpha$, H$\beta$, and \HeI~5876\AA \ emission line luminosities, respectively.}
\par
%\centering

\begin{tabular}{lccccccccccccccccc}
\hline\hline
(1) &(2) &(3) &(4)&(5)&(6)&(7)&(8)&(9)&(10)&(11)&(12)&(13)&(14)&(15)&(16)&(17)&(18)\\
&  RA     & Dec     &              &adopted      & Lum & $EW$(H$_{\alpha}$) &$EW$(Li)   & TTS        & $A_{\rm v}$ &   & &Mass&Age&Disk&Log $\dot{M}_{\rm acc}$&Log $\dot{M}_{\rm acc}$&Log $\dot{M}_{\rm acc}$\\
ID& (J2000) & (J2000) & Spectral type&Spectral type& ($L_{\odot}$) & ($\AA$)       & ($\AA$) &type & (mag) &clu  &$\alpha(3.6{-}8.0~\mu{\rm m})$&($M_{\odot}$) & (Myr) &property &($M_{\odot}$~yr$^{-1}$) &($M_{\odot} $~yr$^{-1}$) & ($M_{\odot} $~yr$^{-1}$) \\ 
\hline
1& 05 35 14.56& --06 15 12.7&M1$^{a}$&M1&0.569&$-$9.8$^{a}$&...&CTTS&2.3&no & $-$1.20 & $\rm 0.42^{+0.06}_{-0.05}$ & $\rm 0.70^{+0.52}_{-0.22}$ & Thick & $-$8.92 & $-$9.11 & $-$8.74\\
2 &  05 35 14.67 &  $-$06 15 07.3 & M5$^{a}$ & M5 & 0.168 & $-$23.2$^{a}$ & ... & CTTS & 2.6 & no & $-$0.75 & $\rm 0.19^{+0.04}_{-0.03}$ & $\rm 0.69^{+0.53}_{-0.24}$ & Thick & $-$9.71 & ... & ...\\
3 &  05 35 14.72 &  $-$06 13 39.9 & M5.5$^{a}$ & M5.5 & 0.292 & $-$19.6$^{a}$ & ... & CTTS & 0.8 & no & $-$2.60 & $\rm 0.15^{+0.03}_{-0.02}$ & $\rm 0.14^{+0.12}_{-0.06}$ & No disk & $-$9.48 & $-$9.86 & $-$9.79\\
4 &  05 35 15.83 &  $-$06 24 45.7 & M5$^{a}$ & M5 & 0.126 & $-$10.9$^{a}$ & 0.9$^{a}$ & WTTS & 0.0 & no & $-$2.56 & $\rm 0.18^{+0.03}_{-0.02}$ & $\rm 0.99^{+0.47}_{-0.27}$ & Thin & ... & ... & ...\\
5 &  05 35 17.80 &  $-$06 24 38.4 & M1$^{a}$, M2$^{b}$ & M1 & 0.964 & $-$6.3$^{a}$, $-$24.0$^{b}$ & 0.7$^{a}$ & WTTS & 0.3 & no & $-$0.35 & $\rm 0.39^{+0.03}_{-0.03}$ & $\rm 0.28^{+0.07}_{-0.06}$ & Thick & ... & ... & ...\\
6 &  05 35 18.26 &  $-$06 24 30.3 & M6$^{a}$, M3.5$^{b}$ & M6 & 0.116 & $-$12.7$^{a}$ & ... & WTTS & 0.0 & no & $-$2.48 & $\rm 0.13^{+0.06}_{-0.02}$ & $\rm 0.51^{+1.06}_{-0.16}$ & Thin & ... & ... & ...\\
7 &  05 35 18.95 &  $-$06 27 25.6 & M4$^{b}$ & M4 & 0.187 & $-$80.0$^{b}$ & ... & CTTS & 0.0 & no & $-$1.16 & $\rm 0.24^{+0.09}_{-0.06}$ & $\rm 1.02^{+1.17}_{-0.47}$ & Thick & $-$8.90 & ... & ...\\
8 &  05 35 19.34 &  $-$06 24 14.5 & M2.5$^{a}$, M2.5$^{b}$ & M2.5 & 0.262 & $-$3.4$^{a}$ & ... & WTTS & 0.0 & no & $-$2.65 & $\rm 0.34^{+0.04}_{-0.03}$ & $\rm 1.34^{+0.48}_{-0.26}$ & No disk & ... & ... & ...\\
9 &  05 35 21.77 &  $-$06 18 51.2 & M4.5$^{a}$ & M4.5 & 0.116 & $-$23.4$^{a}$ & ... & CTTS & 0.0 & no & $-$1.77 & $\rm 0.19^{+0.01}_{-0.01}$ & $\rm 1.30^{+0.23}_{-0.20}$ & Thick & $-$9.98 & $-$10.31 & $-$10.56\\
10 &  05 35 23.60 &  $-$06 28 24.4 & M4$^{a}$, M3.5$^{b}$ & M4 & 0.633 & $-$5.0$^{a}$ & ... & WTTS & 0.0 & no & $-$2.80 & $\rm 0.21^{+0.05}_{-0.03}$ & $\rm 0.11^{+0.09}_{-0.05}$ & No disk & ... & ... & ...\\
11 &  05 35 24.45 &  $-$06 27 47.2 & M4.5$^{a}$, M3.5$^{b}$ & M4.5 & 0.048 & $-$0.2$^{a}$, $-$59.0$^{b}$ & ... & WTTS & 0.0 & no & ... & $\rm 0.21^{+0.06}_{-0.04}$ & $\rm 5.91^{+4.99}_{-2.07}$ & ... & ... & ... & ...\\
12 &  05 35 24.49 &  $-$06 28 40.5 & M5$^{b}$ & M5 & 0.153 & $-$13.0$^{b}$ & ... & WTTS & 0.0 & no & $-$2.55 & $\rm 0.18^{+0.07}_{-0.04}$ & $\rm 0.70^{+1.09}_{-0.32}$ & Thin & ... & ... & ...\\
13 &  05 35 25.11 &  $-$06 47 56.6 & M0$^{b}$ & M0 & 1.146 & ... & ... & ... & 1.1 & no & $-$0.84 & $\rm 0.47^{+0.18}_{-0.09}$ & $\rm 0.35^{+1.48}_{-0.16}$ & Thick & ... & ... & ...\\
14 &  05 35 25.70 &  $-$06 23 18.9 & M3.5$^{a}$, M3$^{b}$ & M3.5 & 0.284 & $-$3.9$^{a}$ & 0.6$^{a}$ & WTTS & 0.0 & no & $-$2.62 & $\rm 0.28^{+0.06}_{-0.04}$ & $\rm 0.75^{+0.64}_{-0.26}$ & No disk & ... & ... & ...\\
15 &  05 35 26.59 &  $-$06 15 32.8 & M3.5$^{a}$ & M3.5 & 0.187 & $-$7.8$^{a}$ & ... & WTTS & 0.0 & no & $-$2.79 & $\rm 0.27^{+0.04}_{-0.03}$ & $\rm 1.27^{+0.85}_{-0.43}$ & No disk & ... & ... & ...\\
16 &  05 35 26.80 &  $-$06 15 34.4 & M4$^{a}$ & M4 & 0.186 & $-$4.9$^{a}$ & ... & WTTS & 0.0 & no & $-$2.70 & $\rm 0.25^{+0.03}_{-0.03}$ & $\rm 1.09^{+0.61}_{-0.32}$ & No disk & ... & ... & ...\\
17 &  05 35 26.83 &  $-$06 26 47.6 & M0$^{a}$, M0$^{b}$ & M0 & 0.567 & $-$2.2$^{a}$ & 0.5$^{a}$ & WTTS & 0.0 & no & $-$2.80 & $\rm 0.59^{+0.20}_{-0.13}$ & $\rm 1.54^{+1.78}_{-0.68}$ & No disk & ... & ... & ...\\
18 &  05 35 26.96 &  $-$06 27 13.0 & M4.5$^{a}$ & M4.5 & 0.050 & $-$6.2$^{a}$ & ... & WTTS & 0.0 & no & ... & $\rm 0.18^{+0.04}_{-0.03}$ & $\rm 4.43^{+2.17}_{-1.22}$ & ... & ... & ... & ...\\
19 &  05 35 27.17 &  $-$06 19 42.0 & M1.5$^{a}$, M2.5$^{b}$ & M1.5 & 1.203 & $-$8.5$^{a}$, $-$10.0$^{b}$ & ... & WTTS & 0.6 & no & $-$1.62 & $\rm 0.35^{+0.04}_{-0.04}$ & $\rm 0.16^{+0.06}_{-0.04}$ & Thick & ... & ... & ...\\
20 &  05 35 27.35 &  $-$06 19 31.3 & M1$^{a}$ & M1 & 0.021 & $-$32.4$^{a}$ & ... & CTTS & 1.1 & no & $-$0.52 & ... & ... & Thick & ... & ... & ...\\
21 &  05 35 28.28 &  $-$06 22 29.3 & M6$^{a}$ & M6 & 0.024 & $-$15.5$^{a}$ & ... & WTTS & 0.0 & no & ... & ... & ... & ... & ... & ... & ...\\
22 &  05 35 29.21 &  $-$06 16 29.7 & M5$^{a}$, M5$^{b}$ & M5 & 0.553 & $-$10.6$^{a}$, $-$14.0$^{b}$ & ... & WTTS & 1.0 & no & $-$2.73 & $\rm 0.19^{+0.02}_{-0.02}$ & $\rm 0.08^{+0.04}_{-0.03}$ & No disk & ... & ... & ...\\
23 &  05 35 29.46 &  $-$06 16 26.6 & M2.5$^{a}$ & M2.5 & 0.500 & $-$3.3$^{a}$ & ... & WTTS & 1.1 & no & $-$1.88 & $\rm 0.30^{+0.03}_{-0.03}$ & $\rm 0.39^{+0.14}_{-0.11}$ & Thin & ... & ... & ...\\
24 &  05 35 30.41 &  $-$06 27 07.2 & M5.5$^{a}$ & M5.5 & 0.018 & $-$11.9$^{a}$ & ... & WTTS & 0.0 & no & ... & $\rm 0.11^{+0.06}_{-0.00}$ & $\rm 6.73^{+14.43}_{-0.48}$ & ... & ... & ... & ...\\
25 &  05 35 31.05 &  $-$06 45 18.1 & M3$^{b}$ & M3 & 0.627 & ... & ... & ... & 0.0 & no & $-$1.02 & $\rm 0.27^{+0.08}_{-0.05}$ & $\rm 0.21^{+0.20}_{-0.11}$ & Thick & ... & ... & ...\\
26 &  05 35 31.49 &  $-$06 14 18.9 & M3$^{a}$ & M3 & 0.193 & $-$10.8$^{a}$ & 0.5$^{a}$ & WTTS & 0.0 & no & $-$2.78 & $\rm 0.31^{+0.04}_{-0.03}$ & $\rm 1.67^{+0.61}_{-0.37}$ & No disk & ... & ... & ...\\
27 &  05 35 31.89 &  $-$06 36 25.5 & M6$^{b}$ & M6 & 0.070 & ... & ... & ... & 0.0 & no & $-$1.17 & $\rm 0.11^{+0.06}_{-0.01}$ & $\rm 0.82^{+2.05}_{-0.10}$ & Thick & ... & ... & ...\\
28 &  05 35 33.93 &  $-$06 14 32.8 & M5$^{a}$ & M5 & 0.235 & $-$50.4$^{a}$ & 0.5$^{a}$ & CTTS & 0.0 & no & $-$1.11 & $\rm 0.19^{+0.01}_{-0.01}$ & $\rm 0.40^{+0.08}_{-0.07}$ & Thick & $-$9.04 & ... & ...\\
29 &  05 35 34.17 &  $-$06 22 57.8 & M3$^{a}$ & M3 & 0.217 & $-$3.2$^{a}$ & 0.4$^{a}$ & WTTS & 0.0 & no & $-$2.61 & $\rm 0.31^{+0.04}_{-0.03}$ & $\rm 1.48^{+0.55}_{-0.34}$ & No disk & ... & ... & ...\\
30 &  05 35 34.19 &  $-$06 39 43.3 & M3.5$^{b}$ & M3.5 & 0.274 & ... & ... & ... & 0.0 & no & $-$1.52 & $\rm 0.27^{+0.09}_{-0.06}$ & $\rm 0.72^{+0.82}_{-0.31}$ & Thick & ... & ... & ...\\
31 &  05 35 34.34 &  $-$06 24 22.8 & M5.5$^{a}$ & M5.5 & 0.032 & $-$17.3$^{a}$ & ... & WTTS & 0.0 & no & $-$1.86 & $\rm 0.13^{+0.04}_{-0.02}$ & $\rm 3.93^{+4.41}_{-0.89}$ & Thin & ... & ... & ...\\
32 &  05 35 38.59 &  $-$06 23 43.1 & M4$^{a}$, M3$^{b}$ & M4 & 0.145 & $-$6.1$^{a}$, $-$13.0$^{b}$ & 0.6$^{a}$ & WTTS & 0.0 & no & $-$2.46 & $\rm 0.26^{+0.04}_{-0.03}$ & $\rm 1.77^{+0.74}_{-0.46}$ & Thin & ... & ... & ...\\
33 &  05 35 40.83 &  $-$06 18 06.7 & M3$^{a}$ & M3 & 0.202 & $-$6.5$^{a}$ & 0.5$^{a}$ & WTTS & 0.0 & no & $-$2.89 & $\rm 0.33^{+0.04}_{-0.03}$ & $\rm 1.79^{+0.71}_{-0.41}$ & No disk & ... & ... & ...\\
34 &  05 35 41.02 &  $-$06 22 45.4 & M0$^{a}$, M1$^{b}$ & M0 & 1.839 & $-$11.6$^{a}$, $-$14.0$^{b}$ & 0.6$^{a}$ & CTTS & 1.0 & no & $-$1.74 & $\rm 0.42^{+0.07}_{-0.05}$ & $\rm 0.13^{+0.08}_{-0.04}$ & Thick & $-$7.84 & $-$8.01 & ...\\
35 &  05 35 41.65 &  $-$06 25 19.2 & M6$^{a}$ & M6 & 0.024 & $-$18.6$^{a}$&...&CTTS&0.5&no&...&...&...&...&...&...&...\\[2pt]
\hline
36 &  05 35 42.09 &  $-$06 16 20.8 & M6$^{a}$ & M6 & 0.027 & $-$21.0$^{a}$ & ... & CTTS & 0.0 & no & $-$1.82 & $\rm 0.10^{+0.05}_{-0.00}$ & $\rm 3.28^{+4.83}_{-0.22}$ & Thin & $-$11.09 & ... & ...\\
37 &  05 35 42.84 &  $-$06 21 44.6 & M2.5$^{a}$ & M2.5 & 0.264 & $-$4.9$^{a}$ & ... & WTTS & 0.0 & no & $-$1.42 & $\rm 0.35^{+0.04}_{-0.03}$ & $\rm 1.36^{+0.54}_{-0.27}$ & Thick & ... & ... & ...\\
38 &  05 35 42.76 &  $-$06 34 51.9 & M0.5$^{b}$ & M0.5 & 0.491 & ... & ... & ... & 0.0 & no & $-$1.06 & $\rm 0.51^{+0.18}_{-0.10}$ & $\rm 1.35^{+2.05}_{-0.50}$ & Thick & ... & ... & ...\\
39 &  05 35 43.37 &  $-$06 22 19.6 & M1.5$^{a}$ & M1.5 & 0.820 & $-$103.4$^{a}$ & 0.3$^{a}$ & CTTS & 2.9 & no & $-$1.55 & $\rm 0.34^{+0.07}_{-0.05}$ & $\rm 0.26^{+2.69}_{-0.10}$ & Thick & $-$7.35 & $-$7.63 & $-$6.97\\
40 &  05 35 44.18 &  $-$06 28 16.4 & M2$^{a}$ & M2 & 0.073 & $-$141.6$^{a}$ & ... & CTTS & 1.2 & no & $-$0.90 & $\rm 0.47^{+0.09}_{-0.16}$ & $\rm 18.64^{+27.19}_{-10.22}$ & Thick & $-$9.23 & $-$9.11 & $-$8.82\\
41 &  05 35 44.27 &  $-$06 26 39.5 & M6$^{a}$ & M6 & 0.035 & $-$17.2$^{a}$ & ... & WTTS & 0.0 & no & ... & ... & ... & ... & ... & ... & ...\\
42 &  05 35 47.65 &  $-$06 21 36.1 & M4$^{b}$ & M4 & 0.293 & $-$30.0$^{b}$ & ... & CTTS & 0.0 & no & $-$1.26 & $\rm 0.23^{+0.07}_{-0.05}$ & $\rm 0.48^{+0.49}_{-0.22}$ & Thick & $-$9.08 & ... & ...\\
43 &  05 35 48.28 &  $-$06 17 59.4 & M2$^{b}$ & M2 & 0.287 & $-$10.0$^{b}$ & ... & WTTS & 0.3 & no & $-$2.73 & $\rm 0.40^{+0.12}_{-0.09}$ & $\rm 1.65^{+1.50}_{-0.72}$ & No disk & ... & ... & ...\\
44 &  05 35 48.39 &  $-$06 17 36.2 & M4$^{b}$ & M4 & 0.214 & $-$11.0$^{b}$ & ... & WTTS & 0.0 & no & $-$2.65 & $\rm 0.24^{+0.09}_{-0.05}$ & $\rm 0.81^{+0.94}_{-0.37}$ & No disk & ... & ... & ...\\
45 &  05 35 50.19 &  $-$06 50 40.2 & K7$^{a}$ & K7 & 0.285 & $-$124.9$^{a}$ & ... & CTTS & 1.3 & no & $-$0.67 & $\rm 0.88^{+0.03}_{-0.21}$ & $\rm 11.83^{+14.86}_{-6.24}$ & Thick & $-$8.19 & ... & ...\\
46 &  05 35 57.30 &  $-$06 40 28.1 & M3.5$^{a}$, M3$^{b}$ & M3.5 & 0.303 & $-$113.8$^{a}$ & 0.4$^{a}$ & CTTS & 0.0 & no & $-$0.94 & $\rm 0.27^{+0.01}_{-0.01}$ & $\rm 0.66^{+0.10}_{-0.08}$ & Thick & $-$8.25 & $-$8.81 & $-$8.95\\
47 &  05 35 57.31 &  $-$06 15 37.7 & M2.5$^{a}$, M3$^{b}$ & M2.5 & 0.284 & $-$34.9$^{a}$, $-$60.0$^{b}$ & 0.4$^{a}$ & CTTS & 0.0 & no & $-$1.46 & $\rm 0.36^{+0.07}_{-0.05}$ & $\rm 1.39^{+0.65}_{-0.43}$ & Thick & $-$8.88 & $-$8.71 & $-$8.62\\
48 &  05 35 57.45 &  $-$06 42 41.9 & K4$^{a}$, M0$^{b}$ & K4 & 3.208 & ... & ... & ... & 2.0 & no & $-$0.62 & $\rm 1.13^{+0.33}_{-0.35}$ & $\rm 0.65^{+1.00}_{-0.35}$ & Thick & ... & ... & ...\\
49 &  05 35 57.45 &  $-$06 16 35.0 & M3$^{a}$, M3$^{b}$ & M3 & 0.900 & $-$1.7$^{a}$ & ... & WTTS & 0.0 & no & $-$2.66 & $\rm 0.25^{+0.03}_{-0.02}$ & $\rm 0.10^{+0.05}_{-0.03}$ & No disk & ... & ... & ...\\
50 &  05 35 57.85 &  $-$06 23 44.4 & M5$^{a}$ & M5 & 0.156 & $-$19.3$^{a}$ & ... & CTTS & 0.0 & no & $-$1.12 & $\rm 0.19^{+0.04}_{-0.03}$ & $\rm 0.79^{+0.48}_{-0.27}$ & Thick & $-$9.87 & $-$10.13 & ...\\
51 &  05 35 58.10 &  $-$06 14 50.9 & M6$^{b}$ & M6 & 0.144 & $-$13.0$^{b}$ & ... & WTTS & 0.0 & no & $-$2.62 & $\rm 0.13^{+0.06}_{-0.02}$ & $\rm 0.32^{+0.61}_{-0.10}$ & No disk & ... & ... & ...\\
52 &  05 35 58.25 &  $-$06 36 43.1 & M0$^{b}$ & M0 & 0.985 & ... & ... & ... & 1.1 & no & $-$1.37 & $\rm 0.48^{+0.18}_{-0.09}$ & $\rm 0.46^{+0.70}_{-0.19}$ & Thick & ... & ... & ...\\
53 &  05 35 58.26 &  $-$06 14 04.6 & M0.5$^{a}$ & M0.5 & 0.105 & $-$2.6$^{a}$ & ... & WTTS & 4.2 & no & $-$1.02 & $\rm 0.64^{+0.05}_{-0.16}$ & $\rm 25.05^{+31.16}_{-13.54}$ & Thick & ... & ... & ...\\
54 &  05 35 59.75 &  $-$06 16 06.4 & K4.5$^{a}$, M0$^{b}$ & K4.5 & 1.336 & $-$4.0$^{a}$, $-$20.0$^{b}$ & 0.4$^{a}$ & WTTS & 0.0 & no & $-$1.49 & $\rm 1.00^{+0.22}_{-0.25}$ & $\rm 1.64^{+10.29}_{-0.77}$ & Thick & ... & ... & ...\\
55 &  05 35 59.84 &  $-$06 42 30.5 & M4.5$^{a}$ & M4.5 & 0.140 & $-$132.2$^{a}$ & ... & CTTS & 0.0 & no & $-$0.97 & $\rm 0.22^{+0.04}_{-0.03}$ & $\rm 1.36^{+0.73}_{-0.35}$ & Thick & $-$8.91 & $-$9.45 & $-$9.08\\
56 &  05 36 00.18 &  $-$06 42 33.9 & K5$^{b}$ & K5 & 3.383 & ... & ... & ... & 2.2 & no & $-$1.46 & $\rm 0.82^{+0.32}_{-0.14}$ & $\rm 0.27^{+0.43}_{-0.10}$ & Thick & ... & $-$7.42 & ...\\
57 &  05 36 01.09 &  $-$06 25 07.7 & M4$^{a}$, M3.5$^{b}$ & M4 & 0.254 & $-$7.9$^{a}$, $-$45.0$^{b}$ & 0.6$^{a}$ & WTTS & 0.0 & no & $-$1.74 & $\rm 0.24^{+0.03}_{-0.02}$ & $\rm 0.61^{+0.31}_{-0.16}$ & Thick & ... & ... & ...\\
58 &  05 36 01.10 &  $-$06 15 31.4 & M1.5$^{a}$, M1$^{b}$ & M1.5 & 0.458 & $-$35.0$^{a}$, $-$120.0$^{b}$ & 0.6$^{a}$ & CTTS & 0.3 & no & $-$1.15 & $\rm 0.43^{+0.08}_{-0.07}$ & $\rm 1.02^{+0.56}_{-0.34}$ & Thick & $-$8.41 & $-$8.41 & $-$8.23\\
59 &  05 36 01.66 &  $-$06 42 36.2 & M5.5$^{a}$, M5$^{b}$ & M5.5 & 0.128 & $-$53.9$^{a}$ & 0.4$^{a}$ & CTTS & 0.1 & no & $-$1.65 & $\rm 0.16^{+0.04}_{-0.03}$ & $\rm 0.74^{+0.57}_{-0.30}$ & Thick & $-$9.50 & $-$10.09 & ...\\
60 &  05 36 05.00 &  $-$06 42 44.0 & M5$^{a}$, M5$^{b}$ & M5 & 0.361 & $-$19.0$^{a}$ & 0.3$^{a}$ & CTTS & 0.0 & no & $-$1.55 & $\rm 0.17^{+0.03}_{-0.02}$ & $\rm 0.14^{+0.08}_{-0.05}$ & Thick & $-$9.27 & $-$9.65 & $-$9.72\\
61 &  05 36 04.98 &  $-$06 46 41.4 & M0.5$^{b}$ & M0.5 & 1.027 & ... & ... & ... & 0.0 & no & $-$1.96 & $\rm 0.43^{+0.13}_{-0.08}$ & $\rm 0.33^{+0.32}_{-0.14}$ & Thin & ... & ... & ...\\
62 &  05 36 05.08 &  $-$06 29 32.2 & M0$^{b}$ & M0 & 0.549 & $-$14.0$^{b}$ & ... & CTTS & 0.0 & no & $-$1.35 & $\rm 0.55^{+0.20}_{-0.11}$ & $\rm 1.37^{+1.75}_{-0.65}$ & Thick & $-$8.76 & ... & ...\\
63 &  05 36 05.16 &  $-$06 25 25.4 & M1.5$^{b}$, M0$^{c}$ & M1.5 & 0.550 & $-$4.8$^{c}$ & 0.6$^{c}$ & WTTS & 0.0 & no & $-$2.75 & $\rm 0.39^{+0.11}_{-0.08}$ & $\rm 0.63^{+0.50}_{-0.27}$ & No disk & ... & ... & ...\\
64 &  05 36 06.59 &  $-$06 31 43.0 & M3.5$^{b}$ & M3.5 & 0.159 & ... & ... & ... & 0.0 & no & $-$1.23 & $\rm 0.28^{+0.11}_{-0.07}$ & $\rm 1.84^{+2.32}_{-0.82}$ & Transition disk & ... & ... & ...\\
65 &  05 36 06.65 &  $-$06 32 17.2 & K5$^{b}$ & K5 & 1.917 & $-$18.0$^{b}$ & ... & CTTS & 4.2 & no & $-$1.38 & $\rm 0.86^{+0.30}_{-0.15}$ & $\rm 0.70^{+1.08}_{-0.27}$ & Thick & $-$7.81 & ... & ...\\
66 &  05 36 06.66 &  $-$06 14 25.9 & K6$^{a}$, M0$^{b}$ & K6 & 1.092 & $-$13.1$^{a}$, $-$22.0$^{b}$ & 0.5$^{a}$ & CTTS & 1.6 & no & $-$1.42 & $\rm 0.78^{+0.18}_{-0.13}$ & $\rm 1.22^{+1.06}_{-0.43}$ & Thick & $-$8.31 & $-$8.01 & $-$7.51\\
67 &  05 36 06.93 &  $-$06 18 53.4 & M4.5$^{b}$ & M4.5 & 0.878 & $-$10.0$^{b}$ & ... & WTTS & 0.0 & no & $-$2.74 & $\rm 0.19^{+0.05}_{-0.01}$ & $\rm 0.03^{+0.06}_{-0.01}$ & No disk & ... & ... & ...\\
68 &  05 36 08.29 &  $-$06 48 36.3 & M0.5$^{a}$, M2$^{b}$ & M0.5 & 0.705 & $-$67.5$^{a}$ & ... & CTTS & 1.7 & no & $-$0.41 & $\rm 0.49^{+0.04}_{-0.04}$ & $\rm 0.75^{+0.23}_{-0.15}$ & Thick & $-$7.69 & $-$7.88 & ...\\
69 &  05 36 08.34 &  $-$06 24 37.9 & M5.5$^{a}$, M6$^{b}$, M7$^{c}$ & M5.5 & 0.252 & $-$14.0$^{a}$, $-$17.0$^{b}$, $-$27.4$^{c}$ & 0.5$^{a}$ & WTTS & 0.0 & yes & $-$1.79 & $\rm 0.15^{+0.03}_{-0.02}$ & $\rm 0.18^{+0.13}_{-0.09}$ & Thick & ... & ... & ...\\
70 &  05 36 08.57 &  $-$06 40 33.6 & M6$^{a}$, M5.5$^{b}$ & M6 & 0.046 & $-$32.8$^{a}$ & 0.5$^{a}$ & CTTS & 0.0 & no & ... & ... & ... & ... & ... & ... & ...\\
71 &  05 36 09.35 &  $-$06 17 10.8 & M3$^{a}$, M2.5$^{b}$ & M3 & 0.483 & $-$3.2$^{a}$ & 0.7$^{a}$ & WTTS & 0.0 & no & $-$2.77 & $\rm 0.30^{+0.03}_{-0.02}$ & $\rm 0.40^{+0.09}_{-0.08}$ & No disk & ... & ... & ...\\
72 &  05 36 09.49 &  $-$06 18 36.3 & M2$^{b}$, M1$^{c}$ & M2 & 0.744 & $-$6.2$^{c}$ & 0.4$^{c}$ & WTTS & 0.0 & no & $-$2.63 & $\rm 0.33^{+0.09}_{-0.06}$ & $\rm 0.27^{+0.21}_{-0.11}$ & No disk & ... & ... & ...\\
73 &  05 36 09.52 &  $-$06 24 33.7 & M4.5$^{b}$, M4.5$^{c}$ & M4.5 & 0.044 & $-$188.0$^{b}$, $-$66.1$^{c}$ & ... & CTTS & 0.0 & yes & $-$1.53 & $\rm 0.20^{+0.11}_{-0.05}$ & $\rm 6.17^{+13.10}_{-2.83}$ & Thick&$-$10.13&...&...\\[2pt]
\hline
74 &  05 36 10.19 &  $-$06 18 54.8 & K7.5$^{a}$, K5$^{b}$ & K7.5 & 0.163 & 0.2$^{a}$, $-$13.0$^{b}$ & ... & CTTS & 3.1 & no & $-$2.47 & $\rm 0.75^{+0.05}_{-0.16}$ & $\rm 23.25^{+27.23}_{-12.47}$ & Thin & $-$9.81 & ... & ...\\
75 &  05 36 10.44 &  $-$06 20 01.5 & K3$^{b}$, K7.5$^{c}$ & K3 & 0.964 & $-$2.3$^{c}$ & 0.3$^{c}$ & WTTS & 0.9 & no & $-$2.76 & $\rm 1.25^{+0.03}_{-0.07}$ & $\rm 6.50^{+3.47}_{-1.85}$ & No disk & ... & ... & ...\\
76 &  05 36 11.12 &  $-$06 18 14.1 & M6$^{a}$, M6$^{c}$ & M6 & 0.043 & $-$19.7$^{a}$, $-$22.4$^{c}$ & 0.8$^{a}$ & CTTS & 0.0 & no & ... & $\rm 0.11^{+0.05}_{-0.01}$ & $\rm 1.89^{+2.67}_{-0.34}$ & ... & $-$10.82 & $-$11.12 & ...\\
77 &  05 36 11.31 &  $-$06 16 56.0 & M5.5$^{a}$ & M5.5 & 0.077 & $-$12.2$^{a}$ & 0.5$^{a}$ & WTTS & 0.0 & no & $-$1.62 & $\rm 0.13^{+0.06}_{-0.02}$ & $\rm 1.03^{+2.32}_{-0.30}$ & Thick & ... & ... & ...\\
78 &  05 36 11.46 &  $-$06 22 22.1 & K7$^{c}$ & K7 & 1.436 & $-$21.6$^{c}$ & ... & CTTS & 6.3 & yes & $-$1.14 & $\rm 0.60^{+0.16}_{-0.14}$ & $\rm 0.44^{+1.54}_{-0.19}$ & Thick & $-$7.76 & ... & ...\\
79 &  05 36 11.67 &  $-$06 24 58.4 & M5$^{a}$, M6.5$^{c}$ & M5 & 0.045 & $-$7.7$^{a}$, $-$11.3$^{c}$ & ... & WTTS & 0.0 & yes & $-$1.23 & $\rm 0.16^{+0.07}_{-0.03}$ & $\rm 3.55^{+5.18}_{-1.33}$ & Transition disk & ... & ... & ...\\
80 &  05 36 12.61 &  $-$06 23 39.6 & M5$^{a}$, M5$^{b}$, M3$^{c}$ & M5 & 0.232 & $-$23.2$^{a}$, $-$36.0$^{b}$, $-$39.3$^{c}$ & ... & CTTS & 0.4 & yes & $-$1.51 & $\rm 0.17^{+0.02}_{-0.01}$ & $\rm 0.31^{+0.10}_{-0.07}$ & Thick & $-$9.50 & $-$10.12 & $-$9.87\\
81 &  05 36 12.97 &  $-$06 23 33.0 & M2.5$^{a}$, M1.5$^{c}$ & M2.5 & 0.910 & $-$4.1$^{a}$, $-$5.0$^{c}$ & 0.6$^{a}$, 0.2$^{c}$ & WTTS & 1.9 & yes & $-$2.67 & $\rm 0.28^{+0.04}_{-0.03}$ & $\rm 0.14^{+0.06}_{-0.05}$ & No disk & ... & ... & ...\\
82 &  05 36 12.97 &  $-$06 23 33.0 & M1.5$^{c}$ & M1.5 & 1.699 & $-$7.3$^{c}$ & ... & WTTS & 3.5 & yes & $-$2.06 & $\rm 0.38^{+0.11}_{-0.06}$ & $\rm 0.12^{+0.46}_{-0.08}$ & Transition disk & ... & ... & ...\\
83 &  05 36 13.15 &  $-$06 25 41.1 & M3$^{a}$, M3$^{b}$, M2.5$^{c}$ & M3 & 0.223 & $-$2.6$^{a}$, $-$3.4$^{c}$ & 0.6$^{a}$, 0.5$^{c}$ & WTTS & 0.0 & yes & $-$2.74 & $\rm 0.32^{+0.04}_{-0.03}$ & $\rm 1.44^{+0.58}_{-0.36}$ & No disk & ... & ... & ...\\
84 &  05 36 13.45 &  $-$06 43 55.1 & M3$^{a}$ & M3 & 0.005 & $-$36.5$^{a}$ & ... & CTTS & 1.8 & no & ... & ... & ... & ... & ... & ... & ...\\
85 &  05 36 14.76 &  $-$06 13 16.9 & M1$^{a}$ & M1 & 0.355 & $-$66.7$^{a}$ & 0.5$^{a}$ & CTTS & 0.2 & no & $-$1.19 & $\rm 0.51^{+0.09}_{-0.08}$ & $\rm 2.09^{+1.02}_{-0.70}$ & Thick & $-$8.29 & $-$8.21 & $-$8.05\\
86 &  05 36 15.07 &  $-$06 17 36.9 & K3$^{a}$, K5$^{b}$ & K3 & 2.356 & $-$2.3$^{a}$, $-$11.0$^{b}$ & 0.4$^{a}$ & WTTS & 1.6 & no & $-$1.19 & $\rm 1.30^{+0.15}_{-0.14}$ & $\rm 1.42^{+0.67}_{-0.35}$ & Thick & ... & ... & ...\\
87 &  05 36 15.60 &  $-$06 27 20.0 & M3.5$^{a}$ & M3.5 & 0.128 & $-$4.4$^{a}$ & 0.5$^{a}$ & WTTS & 0.0 & no & $-$2.72 & $\rm 0.26^{+0.05}_{-0.04}$ & $\rm 2.25^{+1.47}_{-0.70}$ & No disk & ... & ... & ...\\
88 &  05 36 15.84 &  $-$06 14 50.7 & M2.5$^{a}$, M2.5$^{b}$ & M2.5 & 0.214 & $-$106.4$^{a}$, $-$63.0$^{b}$ & 0.6$^{a}$ & CTTS & 0.0 & no & $-$1.50 & $\rm 0.38^{+0.08}_{-0.06}$ & $\rm 2.24^{+2.14}_{-0.84}$ & Transition disk & $-$8.52 & $-$9.38 & $-$9.40\\
89 &  05 36 17.11 &  $-$06 28 20.2 & M5$^{a}$ & M5 & 0.022 & $-$78.5$^{a}$ & ... & CTTS & 1.0 & no & $-$1.44 & $\rm 0.14^{+0.12}_{-0.02}$ & $\rm 8.57^{+31.43}_{-3.57}$ & Thick & $-$10.52 & $-$11.51 & ...\\
90 &  05 36 17.23 &  $-$06 17 24.5 & M5$^{a}$ & M5 & 0.372 & $-$6.4$^{a}$ & 0.6$^{a}$ & WTTS & 0.0 & no & $-$1.61 & $\rm 0.19^{+0.02}_{-0.02}$ & $\rm 0.18^{+0.09}_{-0.06}$ & Transition disk & ... & ... & ...\\
91 &  05 36 18.48 &  $-$06 20 38.7 & M0$^{b}$, M0.5$^{c}$ & M0 & 0.533 & $-$75.0$^{b}$, $-$44.8$^{c}$ & 0.5$^{c}$ & CTTS & 0.6 & yes & $-$1.52 & $\rm 0.56^{+0.21}_{-0.11}$ & $\rm 1.45^{+2.76}_{-0.61}$ & Thick & $-$8.16 & ... & ...\\
92 &  05 36 18.88 &  $-$06 22 04.2 & M3$^{a}$, M3$^{b}$, M2.5$^{c}$ & M3 & 0.130 & $-$7.9$^{a}$, $-$5.3$^{c}$ & ... & WTTS & 0.0 & yes & ... & $\rm 0.32^{+0.03}_{-0.03}$ & $\rm 3.27^{+0.86}_{-0.60}$ & ... & ... & ... & ...\\
93 &  05 36 19.09 &  $-$06 22 50.6 & M6$^{a}$, M7$^{c}$ & M6 & 0.033 & $-$11.6$^{a}$, $-$13.9$^{c}$ & ... & WTTS & 0.0 & yes & ... & $\rm 0.11^{+0.07}_{-0.00}$ & $\rm 2.50^{+6.87}_{-0.54}$ & ... & ... & ... & ...\\
94 &  05 36 19.38 &  $-$06 25 51.3 & M5$^{a}$, M4.5$^{c}$ & M5 & 0.189 & $-$123.7$^{a}$, $-$123.0$^{c}$ & 0.3$^{a}$ & CTTS & 0.4 & yes & $-$1.11 & $\rm 0.19^{+0.03}_{-0.03}$ & $\rm 0.58^{+0.28}_{-0.20}$ & Thick & $-$8.72 & $-$9.05 & $-$9.38\\
95 &  05 36 21.10 &  $-$06 17 17.3 & M4.5$^{a}$, M5$^{b}$ & M4.5 & 0.249 & $-$11.9$^{a}$, $-$14.0$^{b}$ & ... & WTTS & 0.0 & no & $-$2.64 & $\rm 0.20^{+0.01}_{-0.01}$ & $\rm 0.40^{+0.09}_{-0.07}$ & No disk & ... & ... & ...\\
96 &  05 36 21.16 &  $-$06 26 56.9 & M4.5$^{a}$ & M4.5 & 0.049 & $-$22.1$^{a}$ & 0.9$^{a}$ & CTTS & 0.0 & yes & $-$1.46 & $\rm 0.19^{+0.03}_{-0.02}$ & $\rm 4.58^{+1.85}_{-1.09}$ & Thick & $-$10.63 & $-$10.80 & $-$10.41\\
97 &  05 36 21.57 &  $-$06 22 52.4 & M0.5$^{c}$ & M0.5 & 0.636 & $-$70.7$^{c}$ & ... & CTTS & 9.5 & yes & $-$0.83 & $\rm 0.50^{+0.18}_{-0.10}$ & $\rm 0.89^{+1.48}_{-0.36}$ & Thick & $-$7.75 & ... & ...\\
98 &  05 36 21.84 &  $-$06 26 02.0 & M0$^{a}$, K7$^{b}$, K7.5$^{c}$ & M0 & 1.166 & $-$10.2$^{a}$, $-$71.0$^{b}$, $-$18.3$^{c}$ & 0.6$^{a}$, 0.7$^{c}$ & CTTS & 3.8 & yes & $-$1.01 & $\rm 0.49^{+0.08}_{-0.05}$ & $\rm 0.37^{+0.20}_{-0.10}$ & Thick & $-$8.30 & ... & ...\\
99 &  05 36 21.88 &  $-$06 23 29.9 & M1$^{b}$, M4$^{c}$ & M1 & 2.138 & $-$10.0$^{b}$, $-$12.7$^{c}$ & ... & CTTS & 10.2 & yes & 0.14 & $\rm 0.35^{+0.10}_{-0.06}$ & $\rm 0.06^{+3.24}_{-0.03}$ & Thick & $-$7.68 & ... & ...\\
100 &  05 36 21.96 &  $-$06 41 42.0 & M5$^{b}$ & M5 & 0.231 & ... & ... & ... & 0.0 & no & $-$1.43 & $\rm 0.18^{+0.06}_{-0.04}$ & $\rm 0.36^{+0.46}_{-0.18}$ & Thick & ... & ... & ...\\
101 &  05 36 22.47 &  $-$06 23 44.8 & M0.5$^{c}$ & M0.5 & 0.451 & $-$243.3$^{c}$ & 1.3$^{c}$ & CTTS & 6.0 & yes & $-$1.49 & $\rm 0.54^{+0.18}_{-0.11}$ & $\rm 1.70^{+7.39}_{-0.76}$ & Thick & $-$7.38 & ... & ...\\
102 &  05 36 23.26 &  $-$06 19 37.5 & M3.5$^{b}$, M2.5$^{c}$ & M3.5 & 0.156 & $-$8.1$^{c}$ & 0.9$^{c}$ & WTTS & 1.2 & no & $-$2.60 & $\rm 0.28^{+0.11}_{-0.07}$ & $\rm 1.90^{+2.64}_{-0.84}$ & No disk & ... & ... & ...\\
103 &  05 36 24.48 &  $-$06 22 23.2 & M4$^{a}$, M3$^{c}$ & M4 & 0.380 & $-$8.5$^{a}$, $-$16.5$^{c}$ & 0.6$^{a}$ & WTTS & 2.5 & yes & $-$2.06 & $\rm 0.23^{+0.03}_{-0.03}$ & $\rm 0.30^{+0.22}_{-0.11}$ & Thin & ... & ... & ...\\
104 &  05 36 24.54 &  $-$06 52 34.2 & M5$^{a}$ & M5 & 0.132 & $-$50.0$^{a}$ & 0.5$^{a}$ & CTTS & 0.0 & no & $-$1.51 & $\rm 0.19^{+0.02}_{-0.02}$ & $\rm 0.99^{+0.31}_{-0.21}$ & Thick & $-$9.48 & $-$9.67 & $-$9.64\\
105 &  05 36 25.13 &  $-$06 44 41.9 & K4.5$^{a}$, K3$^{b}$ & K4.5 & 3.575 & $-$8.7$^{a}$ & 0.4$^{a}$ & CTTS & 6.0 & yes & $-$0.34 & $\rm 0.92^{+0.40}_{-0.25}$ & $\rm 0.34^{+0.70}_{-0.17}$ & Thick & $-$7.73 & ... & ...\\
106 &  05 36 25.28 &  $-$06 23 07.2 & M7$^{c}$ & M7 & 0.041 & $-$13.7$^{c}$ & 0.9$^{c}$ & WTTS & 0.0 & yes & ... & ... & ... & ... & ... & ... & ...\\
107 &  05 36 25.41 &  $-$06 24 31.2 & M6$^{a}$, M6$^{c}$ & M6 & 0.083 & $-$9.8$^{a}$, $-$17.6$^{c}$ & ... & WTTS & 0.0 & yes & $-$2.70 & $\rm 0.12^{+0.06}_{-0.01}$ & $\rm 0.71^{+1.49}_{-0.15}$ & No disk & ... & ... & ...\\
108 &  05 36 25.55 &  $-$06 51 28.3 & M5$^{a}$ & M5 & 0.112 & $-$10.2$^{a}$ & 0.5$^{a}$ & WTTS & 0.0 & no & $-$2.53 & $\rm 0.16^{+0.04}_{-0.03}$ & $\rm 0.93^{+0.80}_{-0.34}$ & Thin & ... & ... & ...\\
109 &  05 36 26.08 &  $-$06 26 15.5 & M3.5$^{b}$, M8.5$^{c}$ & M3.5 & 0.094 & $-$12.4$^{a}$, $-$10.0$^{b}$, $-$74.2$^{c}$ & ... & WTTS & 2.9 & yes & $-$2.72 & $\rm 0.29^{+0.13}_{-0.07}$ & $\rm 4.34^{+6.75}_{-1.97}$ & No disk & ... & ... & ...\\
110 &  05 36 26.71 &  $-$06 26 29.0 & M0.5$^{b}$, K7.5$^{c}$ & M0.5 & 1.866 & $-$3.2$^{c}$ & 0.4$^{c}$ & WTTS & 1.1 & yes & $-$2.66 & $\rm 0.39^{+0.12}_{-0.07}$ & $\rm 0.11^{+0.14}_{-0.05}$ & No disk & ...&...&...\\[2pt]
\hline
111 &  05 36 26.83 &  $-$06 24 57.5 & M5$^{c}$ & M5 & 0.196 & $-$16.9$^{c}$ & 0.6$^{c}$ & WTTS & 0.0 & yes & $-$2.81 & $\rm 0.17^{+0.06}_{-0.04}$ & $\rm 0.42^{+0.63}_{-0.19}$ & No disk & ... & ... & ...\\
112 &  05 36 27.01 &  $-$06 21 07.6 & M3$^{b}$, M2.5$^{c}$ & M3 & 0.023 & $-$15.0$^{b}$ & ... & CTTS & 0.0 & yes & ... & $\rm 0.36^{+0.04}_{-0.11}$ & $\rm 59.50^{+24.98}_{-31.27}$ & ... & ... & ... & ...\\
113 &  05 36 27.72 &  $-$06 23 12.3 & M5$^{c}$ & M5 & 0.570 & $-$36.5$^{c}$ & ... & CTTS & 0.3 & yes & $-$1.23 & $\rm 0.17^{+0.05}_{-0.01}$ & $\rm 0.04^{+0.09}_{-0.02}$ & Thick & $-$8.56 & ... & ...\\
114 &  05 36 27.89 &  $-$06 25 36.0 & M3$^{c}$ & M3 & 0.230 & $-$21.9$^{c}$ & 0.2$^{c}$ & CTTS & 0.1 & yes & $-$1.91 & $\rm 0.30^{+0.10}_{-0.07}$ & $\rm 1.26^{+1.31}_{-0.59}$ & Transition disk & $-$9.35 & ... & ...\\
115 &  05 36 28.10 &  $-$06 44 32.6 & M2.5$^{b}$ & M2.5 & 0.252 & ... & ... & ... & 0.4 & yes & $-$1.87 & $\rm 0.35^{+0.12}_{-0.08}$ & $\rm 1.54^{+1.71}_{-0.70}$ & Thin & ... & ... & ...\\
116 &  05 36 29.06 &  $-$06 38 40.7 & M8$^{a}$, M3$^{b}$ & M8 & 0.053 & $-$42.0$^{a}$ & ... & CTTS & 0.0 & no & $-$1.49 & ... & ... & Thick & ... & ... & ...\\
117 &  05 36 29.59 &  $-$06 38 50.4 & M4.5$^{a}$, M5$^{b}$ & M4.5 & 0.109 & $-$8.2$^{a}$ & 0.3$^{a}$ & WTTS & 0.0 & no & ... & $\rm 0.19^{+0.04}_{-0.03}$ & $\rm 1.43^{+0.76}_{-0.44}$ & ... & ... & ... & ...\\
118 &  05 36 30.10 &  $-$06 23 10.2 & M0.5$^{b}$, K6$^{c}$ & M0.5 & 0.450 & $-$13.0$^{b}$, $-$7.3$^{c}$ & 0.5$^{c}$ & WTTS & 0.9 & yes & $-$1.16 & $\rm 0.52^{+0.18}_{-0.11}$ & $\rm 1.58^{+2.59}_{-0.69}$ & Thick & ... & ... & ...\\
119 &  05 36 30.23 &  $-$06 42 46.1 & M1.5$^{a}$, M2$^{b}$ & M1.5 & 0.664 & $-$35.8$^{a}$ & ... & CTTS & 6.5 & yes & $-$1.08 & $\rm 0.37^{+0.10}_{-0.06}$ & $\rm 0.42^{+5.19}_{-0.16}$ & Thick & $-$8.09 & ... & ...\\
120 &  05 36 30.50 &  $-$06 23 56.6 & M4.5$^{b}$, M6$^{c}$ & M4.5 & 0.028 & $-$36.1$^{c}$ & ... & CTTS & 0.0 & yes & $-$1.75 & $\rm 0.19^{+0.10}_{-0.05}$ & $\rm 11.35^{+20.32}_{-5.38}$ & Thick & $-$10.78 & ... & ...\\
121 &  05 36 30.52 &  $-$06 42 03.2 & M6$^{a}$, M6$^{b}$ & M6 & 0.030 & $-$21.1$^{a}$ & 0.6$^{a}$ & CTTS & 0.0 & yes & ... & $\rm 0.10^{+0.06}_{-0.00}$ & $\rm 2.56^{+6.39}_{-0.01}$ & ... & $-$11.00 & $-$11.13 & $-$11.30\\
122 &  05 36 30.96 &  $-$06 52 41.0 & M2$^{a}$ & M2 & 0.012 & ... & ... & ... & 1.7 & no & 0.40 & ... & ... & Thick & ... & ... & ...\\
123 &  05 36 31.84 &  $-$06 23 23.1 & M6.5$^{c}$ & M6.5 & 0.040 & $-$37.0$^{c}$ & ... & CTTS & 0.0 & yes & $-$1.27 & $\rm 0.12^{+0.07}_{-0.01}$ & $\rm 2.63^{+6.52}_{-0.59}$ & Thick & $-$10.55 & ... & ...\\
124 &  05 36 32.38 &  $-$06 19 19.9 & G8$^{b}$ & G8 & 4.874 & ... & ... & ... & 2.9 & no & $-$2.03 & $\rm 1.96^{+0.07}_{-0.10}$ & $\rm 3.67^{+0.99}_{-0.73}$ & Transition disk & ... & ... & ...\\
125 &  05 36 32.42 &  $-$06 40 43.0 & M5$^{a}$ & M5 & 0.144 & $-$39.2$^{a}$ & 0.4$^{a}$ & CTTS & 0.0 & no & $-$1.25 & $\rm 0.19^{+0.02}_{-0.02}$ & $\rm 0.90^{+0.29}_{-0.22}$ & Thick & $-$9.54 & $-$9.83 & $-$9.88\\
126 &  05 36 32.82 &  $-$06 39 54.9 & K7.5$^{a}$, M0$^{b}$ & K7.5 & 0.442 & $-$0.7$^{a}$ & 0.3$^{a}$ & WTTS & 0.3 & no & $-$2.99 & $\rm 0.69^{+0.16}_{-0.14}$ & $\rm 3.10^{+2.61}_{-1.27}$ & No disk & ... & ... & ...\\
127 &  05 36 32.89 &  $-$06 44 20.9 & M1$^{a}$, M2$^{b}$ & M1 & 0.544 & $-$70.7$^{a}$ & ... & CTTS & 4.8 & yes & $-$1.04 & $\rm 0.43^{+0.20}_{-0.12}$ & $\rm 0.81^{+10.19}_{-0.47}$ & Thick & $-$7.88 & $-$8.13 & $-$7.95\\
128 &  05 36 33.81 &  $-$06 19 27.0 & M5$^{c}$ & M5 & 0.155 & $-$9.7$^{c}$ & 1.1$^{c}$ & WTTS & 0.0 & no & $-$2.72 & $\rm 0.17^{+0.07}_{-0.04}$ & $\rm 0.60^{+0.96}_{-0.27}$ & No disk & ... & ... & ...\\
129 &  05 36 35.23 &  $-$06 50 12.2 & M6$^{a}$, M6$^{b}$ & M6 & 0.058 & $-$5.1$^{a}$ & 0.5$^{a}$ & WTTS & 0.0 & no & ... & $\rm 0.11^{+0.06}_{-0.01}$ & $\rm 1.11^{+2.54}_{-0.14}$ & ... & ... & ... & ...\\
130 &  05 36 35.76 &  $-$06 42 49.9 & M2.5$^{a}$ & M2.5 & 0.213 & $-$34.1$^{a}$ & ... & CTTS & 2.5 & yes & $-$1.10 & $\rm 0.34^{+0.14}_{-0.09}$ & $\rm 1.87^{+6.00}_{-0.99}$ & Thick & $-$9.18 & $-$9.40 & $-$9.24\\
131 &  05 36 36.93 &  $-$06 33 24.2 & K8$^{b}$ & K8 & 1.913 & $-$110.0$^{b}$ & ... & CTTS & 3.9 & no & $-$1.27 & $\rm 0.53^{+0.14}_{-0.11}$ & $\rm 0.22^{+0.24}_{-0.10}$ & Thick & $-$6.63 & ... & ...\\
132 &  05 36 40.69 &  $-$06 30 09.3 & M0$^{b}$ & M0 & 0.570 & ... & ... & ... & 0.0 & no & $-$1.57 & $\rm 0.55^{+0.20}_{-0.11}$ & $\rm 1.28^{+1.70}_{-0.57}$ & Thick & ... & ... & ...\\
133 &  05 36 40.69 &  $-$06 52 04.5 & M2$^{a}$ & M2 & 0.383 & $-$6.2$^{a}$ & 0.5$^{a}$ & WTTS & 0.0 & no & $-$2.64 & $\rm 0.37^{+0.06}_{-0.05}$ & $\rm 0.90^{+0.36}_{-0.26}$ & No disk & ... & ... & ...\\
134 &  05 36 40.94 &  $-$06 41 17.8 & M5$^{a}$ & M5 & 0.204 & $-$20.1$^{a}$ & 0.5$^{a}$ & CTTS & 0.5 & no & $-$1.23 & $\rm 0.19^{+0.02}_{-0.02}$ & $\rm 0.54^{+0.16}_{-0.12}$ & Thick & $-$9.64 & $-$10.15 & $-$10.00\\
135 &  05 36 41.35 &  $-$06 34 00.4 & M0$^{b}$ & M0 & 4.188 & $-$148.0$^{b}$ & ... & CTTS & 5.7 & no & $-$0.30 & $\rm 0.41^{+0.16}_{-0.06}$ & $\rm 0.03^{+0.06}_{-0.02}$ & Thick & $-$5.82 & ... & ...\\
136 &  05 36 41.98 &  $-$06 20 45.4 & M6$^{b}$ & M6 & 0.056 & $-$17.0$^{b}$ & ... & WTTS & 0.0 & no & $-$2.72 & $\rm 0.11^{+0.06}_{-0.00}$ & $\rm 1.08^{+2.41}_{-0.08}$ & No disk & ... & ... & ...\\
137 &  05 36 43.77 &  $-$06 15 28.6 & M4$^{b}$ & M4 & 0.215 & $-$10.0$^{b}$ & ... & WTTS & 0.0 & no & $-$1.36 & $\rm 0.24^{+0.09}_{-0.05}$ & $\rm 0.81^{+0.96}_{-0.36}$ & Thick & ... & ... & ...\\
138 &  05 36 45.21 &  $-$06 28 09.6 & M3.5$^{b}$ & M3.5 & 0.453 & $-$12.0$^{b}$ & ... & WTTS & 0.0 & no & $-$2.66 & $\rm 0.25^{+0.07}_{-0.05}$ & $\rm 0.29^{+0.26}_{-0.14}$ & No disk & ... & ... & ...\\
139 &  05 36 45.79 &  $-$06 48 16.0 & M4.5$^{a}$ & M4.5 & 0.007 & $-$4.6$^{a}$ & ... & WTTS & 0.0 & no & ... & $\rm 0.16^{+0.02}_{-0.03}$ & $\rm 65.63^{+21.97}_{-24.53}$ & ... & ... & ... & ...\\
140 &  05 36 46.62 &  $-$06 39 50.0 & M5.5$^{a}$, M2$^{b}$ & M5.5 & 0.065 & $-$15.2$^{a}$ & 0.4$^{a}$ & WTTS & 0.0 & no & ... & $\rm 0.14^{+0.03}_{-0.02}$ & $\rm 1.53^{+1.13}_{-0.44}$ & ... & ... & ... & ...\\
141 &  05 36 50.14 &  $-$06 41 29.2 & K5$^{a}$, M0$^{b}$ & K5 & 1.849 & $-$7.6$^{a}$ & 0.5$^{a}$ & CTTS & 3.5 & no & $-$1.68 & $\rm 0.85^{+0.31}_{-0.18}$ & $\rm 0.72^{+1.22}_{-0.30}$ & Thick & $-$8.30 & $-$7.96 & $-$7.34\\
142 &  05 36 50.23 &  $-$06 48 58.2 & M4.5$^{a}$, M5$^{b}$ & M4.5 & 0.333 & $-$5.3$^{a}$ & 0.6$^{a}$ & WTTS & 0.0 & no & $-$2.57 & $\rm 0.20^{+0.01}_{-0.01}$ & $\rm 0.25^{+0.04}_{-0.04}$ & No disk & ... & ... & ...\\
143 &  05 36 54.41 &  $-$06 51 05.5 & M5.5$^{a}$ & M5.5 & 0.019 & $-$1.8$^{a}$ & ... & WTTS & 0.0 & no & ... & $\rm 0.11^{+0.10}_{-0.01}$ & $\rm 6.56^{+24.94}_{-0.88}$ & ... & ... & ... & ...\\
144 &  05 37 49.54 &  $-$06 56 27.4 & M3.5$^{a}$ & M3.5 & 0.586 & $-$20.0$^{a}$ & 0.6$^{a}$ & CTTS & 0.6 & yes & $-$1.98 & $\rm 0.26^{+0.03}_{-0.03}$ & $\rm 0.20^{+0.08}_{-0.06}$ & Transition disk & $-$8.65 & $-$8.88 & $-$8.75\\
145 &  05 37 51.72 &  $-$06 56 51.9 & M1.5$^{a}$ & M1.5 & 1.324 & $-$39.0$^{a}$ & 0.5$^{a}$ & CTTS & 1.4 & yes & $-$1.43 & $\rm 0.34^{+0.08}_{-0.05}$ & $\rm 0.12^{+0.15}_{-0.05}$ & Thick & $-$7.47 & $-$7.38 & $-$7.02\\
146 &  05 37 53.28 &  $-$07 02 27.2 & M4.5$^{a}$ & M4.5 & 0.059 & $-$3.7$^{a}$ & ... & WTTS & 0.0 & no & ... & $\rm 0.20^{+0.05}_{-0.04}$ & $\rm 4.13^{+3.06}_{-1.44}$ & ... & ... & ... & ...\\
147 &  05 37 54.48 &  $-$06 57 31.1 & K6.5$^{a}$ & K6.5 & 0.939 & $-$2.6$^{a}$ & 0.4$^{a}$ & WTTS & 0.9 & yes & $-$1.11 & $\rm 0.72^{+0.28}_{-0.24}$ & $\rm 1.21^{+4.48}_{-0.70}$ & Thick & ... & ... & ...\\[2pt]
\hline
148 &  05 37 54.51& $-$06 56 45.5&K7$^{a}$&K7&2.799&$-$0.5$^{a}$ & 0.5$^{a}$ & WTTS & 1.9 & yes & $-$2.65 & $\rm 0.57^{+0.10}_{-0.10}$ & $\rm 0.15^{+0.10}_{-0.05}$ & No disk & ... & ... & ...\\
149 &  05 37 55.15 &  $-$06 57 40.7 & M5.5$^{a}$ & M5.5 & 0.135 & $-$14.5$^{a}$ & 0.6$^{a}$ & WTTS & 0.2 & yes & ... & $\rm 0.16^{+0.05}_{-0.03}$ & $\rm 0.62^{+0.70}_{-0.26}$ & ... & ... & ... & ...\\
150 &  05 37 55.22 &  $-$06 57 35.8 & M4.5$^{a}$ & M4.5 & 0.273 & $-$6.3$^{a}$ & 0.6$^{a}$ & WTTS & 0.2 & yes & $-$1.80 & $\rm 0.20^{+0.03}_{-0.02}$ & $\rm 0.35^{+0.15}_{-0.11}$ & Thin & ... & ... & ...\\
151 &  05 37 55.65 &  $-$06 57 18.1 & M0$^{a}$ & M0 & 1.554 & $-$1.2$^{a}$ & ... & WTTS & 2.1 & yes & $-$2.58 & $\rm 0.44^{+0.23}_{-0.10}$ & $\rm 0.19^{+0.49}_{-0.10}$ & No disk & ... & ... & ...\\
152 &  05 37 55.74 &  $-$06 58 37.9 & M3$^{a}$ & M3 & 0.243 & $-$3.7$^{a}$ & ... & WTTS & 1.3 & yes & $-$1.00 & $\rm 0.30^{+0.06}_{-0.04}$ & $\rm 1.10^{+0.78}_{-0.35}$ & Thick & ... & ... & ...\\
153 &  05 37 55.99 &  $-$06 52 33.7 & M5.5$^{a}$ & M5.5 & 0.161 & $-$11.8$^{a}$ & ... & WTTS & 0.0 & no & $-$2.66 & $\rm 0.15^{+0.02}_{-0.02}$ & $\rm 0.42^{+0.23}_{-0.14}$ & No disk & ... & ... & ...\\
154 &  05 38 00.67 &  $-$07 01 14.3 & M4$^{a}$ & M4 & 0.021 & $-$5.6$^{a}$ & ... & WTTS & 0.0 & no & ... & $\rm 0.25^{+0.09}_{-0.07}$ & $\rm 29.50^{+33.28}_{-14.72}$ & ... & ... & ... & ...\\
155 &  05 38 04.84 &  $-$07 02 21.6 & M5.5$^{a}$ & M5.5 & 0.136 & $-$12.8$^{a}$ & ... & WTTS & 2.7 & no & $-$1.12 & $\rm 0.15^{+0.03}_{-0.03}$ & $\rm 0.55^{+0.57}_{-0.20}$ & Thick & ... & ... & ...\\
156 &  05 38 10.50 &  $-$06 57 07.1 & M3$^{a}$ & M3 & 0.194 & $-$9.7$^{a}$ & 0.6$^{a}$ & WTTS & 1.5 & no & $-$1.71 & $\rm 0.34^{+0.05}_{-0.04}$ & $\rm 2.10^{+1.08}_{-0.67}$ & Transition disk & ... & ... & ...\\
157 &  05 38 13.45 &  $-$07 06 43.3 & K4.5$^{a}$ & K4.5 & 2.519 & $-$5.4$^{a}$ & ... & WTTS & 10.3 & no & $-$0.99 & $\rm 0.92^{+0.45}_{-0.34}$ & $\rm 0.56^{+7.23}_{-0.35}$ & Thick & ... & ... & ...\\
158 &  05 38 19.52 &  $-$06 55 29.7 & M3$^{a}$ & M3 & 0.382 & $-$2.7$^{a}$ & ... & WTTS & 7.1 & no & $-$2.73 & $\rm 0.30^{+0.11}_{-0.07}$ & $\rm 0.60^{+0.75}_{-0.27}$ & No disk & ... & ... & ...\\
159 &  05 38 21.22 &  $-$07 01 20.3 & M5$^{a}$ & M5 & 0.102 & $-$47.5$^{a}$ & ... & CTTS & 1.4 & no & $-$1.27 & $\rm 0.17^{+0.03}_{-0.03}$ & $\rm 1.25^{+0.75}_{-0.38}$ & Thick & $-$9.70 & $-$10.04 & $-$9.59\\
160 &  05 38 34.06 &  $-$07 04 36.8 & M4$^{a}$ & M4 & 0.007 & $-$1.3$^{a}$ & ... & WTTS & 0.0 & no & ... & $\rm 0.20^{+0.02}_{-0.06}$ & $\rm 101.66^{+4.08}_{-51.84}$ & ... & ... & ... & ...\\
161 &  05 38 40.22 &  $-$06 56 53.3 & M2$^{a}$ & M2 & 0.208 & $-$3.1$^{a}$ & ... & WTTS & 2.4 & yes & $-$2.63 & $\rm 0.39^{+0.13}_{-0.09}$ & $\rm 2.61^{+9.58}_{-1.34}$ & No disk & ... & ... & ...\\
162 &  05 38 40.59 &  $-$07 02 23.8 & M3$^{a}$ & M3 & 0.155 & $-$107.1$^{a}$ & ... & CTTS & 3.5 & yes & $-$1.11 & $\rm 0.32^{+0.23}_{-0.12}$ & $\rm 2.59^{+19.02}_{-1.68}$ & Thick & $-$8.81 & $-$9.65 & $-$9.33\\
163 &  05 38 41.48 &  $-$07 01 52.6 & M4.5$^{a}$ & M4.5 & 0.183 & $-$9.6$^{a}$ & 0.5$^{a}$ & WTTS & 2.6 & yes & $-$1.15 & $\rm 0.21^{+0.05}_{-0.04}$ & $\rm 0.78^{+0.65}_{-0.36}$ & Thick & ... & ... & ...\\
164 &  05 38 41.56 &  $-$06 52 51.0 & M5$^{a}$ & M5 & 0.140 & $-$9.4$^{a}$ & 0.6$^{a}$ & WTTS & 0.2 & no & $-$2.68 & $\rm 0.19^{+0.06}_{-0.04}$ & $\rm 0.91^{+0.77}_{-0.38}$ & No disk & ... & ... & ...\\
165 &  05 38 43.22 &  $-$06 58 08.9 & K7.5$^{a}$ & K7.5 & 14.069 & $-$98.7$^{a}$ & ... & CTTS & 12.6 & yes & $-$1.27 & ... & ... & Thick & ... & ... & ...\\
166 &  05 38 43.84 &  $-$06 58 22.3 & M2.5$^{a}$ & M2.5 & 0.359 & $-$66.9$^{a}$ & ... & CTTS & 4.5 & yes & $-$0.52 & $\rm 0.33^{+0.15}_{-0.08}$ & $\rm 0.80^{+6.68}_{-0.48}$ & Thick & $-$8.36 & $-$9.13 & $-$9.00\\
167 &  05 38 44.95 &  $-$06 58 14.7 & M1$^{a}$ & M1 & 0.598 & $-$37.3$^{a}$ & ... & CTTS & 6.0 & yes & $-$0.97 & $\rm 0.45^{+0.03}_{-0.03}$ & $\rm 0.77^{+0.17}_{-0.14}$ & Thick & $-$8.15 & $-$7.22 & $-$7.56\\
168 &  05 38 46.84 &  $-$07 05 08.9 & M2.5$^{a}$ & M2.5 & 0.243 & $-$99.3$^{a}$ & ... & CTTS & 4.3 & yes & $-$1.64 & $\rm 0.34^{+0.07}_{-0.06}$ & $\rm 1.52^{+1.38}_{-0.62}$ & Thick & $-$8.49 & $-$9.19 & $-$9.14\\
169 &  05 38 47.17 &  $-$07 02 40.4 & K7$^{a}$ & K7 & 6.562 & $-$2.3$^{a}$ & ... & WTTS & 13.0 & yes & $-$2.69 & $\rm 0.53^{+0.76}_{-0.14}$ & $\rm 0.03^{+2.60}_{-0.02}$ & No disk & ... & ... & ...\\
170 &  05 38 47.74 &  $-$07 06 14.9 & M1$^{a}$ & M1 & 1.473 & $-$76.8$^{a}$ & ... & CTTS & 10.4 & no & $-$1.19 & $\rm 0.37^{+0.31}_{-0.10}$ & $\rm 0.13^{+6.60}_{-0.08}$ & Thick & $-$7.00 & ... & ...\\
171 &  05 38 47.93 &  $-$07 05 06.1 & M5$^{a}$ & M5 & 0.378 & $-$10.3$^{a}$ & 0.5$^{a}$ & WTTS & 4.1 & yes & $-$2.62 & $\rm 0.18^{+0.03}_{-0.02}$ & $\rm 0.16^{+1.16}_{-0.06}$ & No disk & ... & ... & ...\\
172 &  05 38 49.88 &  $-$07 02 35.4 & M0$^{a}$ & M0 & 0.087 & $-$4.4$^{a}$ & ... & WTTS & 6.6 & yes & ... & $\rm 0.63^{+0.08}_{-0.30}$ & $\rm 44.51^{+26.80}_{-33.82}$ & ... & ... & ... & ...\\
173 &  05 38 50.44 &  $-$07 00 43.0 & K6$^{a}$ & K6 & 4.113 & $-$1.5$^{a}$ & ... & WTTS & 10.9 & yes & $-$2.35 & $\rm 0.65^{+0.56}_{-0.22}$ & $\rm 0.12^{+3.62}_{-0.07}$ & Thin & ... & ... & ...\\
174 &  05 38 52.97 &  $-$07 05 50.6 & M6$^{a}$ & M6 & 0.055 & $-$32.0$^{a}$ & ... & CTTS & 0.0 & no & $-$2.56 & $\rm 0.10^{+0.07}_{-0.00}$ & $\rm 0.97^{+2.95}_{-0.10}$ & No disk & $-$10.35 & ... & ...\\
175 &  05 38 53.33 &  $-$07 05 45.2 & K3.5$^{a}$ & K3.5 & 0.016 & $-$30.9$^{a}$ & ... & CTTS & 2.1 & no & 0.47 & ... & ... & Thick & ... & ... & ...\\
176 &  05 38 54.45 &  $-$07 04 46.7 & M3$^{a}$ & M3 & 0.177 & $-$7.7$^{a}$ & 0.6$^{a}$ & WTTS & 2.6 & yes & $-$2.72 & $\rm 0.33^{+0.04}_{-0.03}$ & $\rm 2.18^{+1.45}_{-0.52}$ & No disk & ... & ... & ...\\
177 &  05 38 54.93 &  $-$07 07 10.9 & M3.5$^{a}$ & M3.5 & 0.156 & $-$12.1$^{a}$ & 0.7$^{a}$ & WTTS & 2.1 & no & $-$1.30 & $\rm 0.30^{+0.03}_{-0.02}$ & $\rm 2.16^{+0.67}_{-0.56}$ & Thick & ... & ... & ...\\
178 &  05 38 55.02 &  $-$06 56 18.6 & F5$^{a}$ & F5 & 0.120 & $-$0.5$^{a}$ & ... & WTTS & 5.8 & no & 0.01 & ... & ... & Thick & ... & ... & ...\\
179 &  05 38 57.04 &  $-$06 52 59.3 & M1.5$^{a}$ & M1.5 & 0.156 & $-$184.8$^{a}$ & 0.3$^{a}$ & CTTS & 0.7 & no & $-$1.33 & $\rm 0.55^{+0.09}_{-0.10}$ & $\rm 8.06^{+5.66}_{-2.99}$ & Thick & $-$8.44 & $-$8.76 & $-$8.34\\
180 &  05 40 19.41 &  $-$08 14 16.4 & K7$^{b}$ & K7 & 1.152 & ... & ... & ... & 5.8 & no & $-$0.67 & $\rm 0.63^{+0.16}_{-0.15}$ & $\rm 0.68^{+0.75}_{-0.29}$ & Thick & ... & ... & ...\\
181 &  05 40 21.84 &  $-$08 08 55.9 & M3$^{b}$ & M3 & 0.269 & $-$25.0$^{b}$ & ... & CTTS & 2.4 & no & $-$1.24 & $\rm 0.31^{+0.11}_{-0.07}$ & $\rm 1.01^{+2.93}_{-0.53}$ & Thick & $-$9.15 & ... & ...\\
182 &  05 40 24.97 &  $-$07 55 35.3 & M2$^{a}$ & M2 & 1.723 & $-$77.5$^{a}$ & ... & CTTS & 8.2 & no & $-$0.91 & $\rm 0.28^{+0.10}_{-0.04}$ & $\rm 0.05^{+0.08}_{-0.03}$ & Thick & $-$6.94 & $-$7.63 & $-$6.88\\
183 &  05 40 24.97 &  $-$08 07 33.2 & K0$^{b}$ & K0 & 8.857 & $-$47.0$^{b}$ & ... & CTTS & 4.4 & no & $-$0.53 & $\rm 2.43^{+0.09}_{-0.27}$ & $\rm 1.05^{+0.64}_{-0.35}$ & Thick & $-$6.54 & ... & ...\\[2pt]
\hline
184 &  05 40 24.97 &  $-$07 55 35.3 & M3$^{b}$ & M3 & 1.557 & ... & ... & ... & 8.0 & no & $-$0.91 & $\rm 0.24^{+0.06}_{-0.02}$ & $\rm 0.03^{+0.05}_{-0.02}$ & Thick & ... & ... & ...\\
185 &  05 40 25.75 &  $-$08 11 16.8 & M3$^{b}$ & M3 & 0.385 & $-$38.0$^{b}$ & ... & CTTS & 4.2 & no & $-$1.16 & $\rm 0.29^{+0.09}_{-0.06}$ & $\rm 0.53^{+0.63}_{-0.23}$ & Thick & $-$8.63 & ... & ...\\
186 &  05 40 27.14 &  $-$08 07 36.5 & M3$^{b}$ & M3 & 0.184 & $-$230.0$^{b}$ & ... & CTTS & 1.5 & no & $-$0.80 & $\rm 0.32^{+0.12}_{-0.08}$ & $\rm 1.98^{+6.20}_{-1.02}$ & Thick & $-$8.26 & ... & ...\\
187 &  05 40 35.44 &  $-$07 56 36.5 & M0.5$^{a}$ & M0.5 & 0.596 & $-$2.3$^{a}$ & ... & WTTS & 8.5 & no & $-$2.76 & $\rm 0.51^{+0.38}_{-0.21}$ & $\rm 1.04^{+11.36}_{-0.71}$ & No disk & ... & ... & ...\\
188 &  05 40 37.36 &  $-$08 04 03.0 & F$^{b}$ & F & 60.687 & ... & ... & ... & 6.7 & no & $-$0.82 & $\rm 3.01^{+0.18}_{-0.06}$ & $\rm 1.94^{+0.16}_{-0.30}$ & Thick & ... & ... & ...\\
189 &  05 40 41.01 &  $-$08 02 18.6 & M0$^{b}$ & M0 & 1.126 & ... & ... & ... & 7.9 & no & $-$1.16 & $\rm 0.47^{+0.17}_{-0.09}$ & $\rm 0.36^{+0.51}_{-0.14}$ & Thick & ... & ... & ...\\
190 &  05 40 42.66 &  $-$07 46 03.0 & M5.5$^{a}$ & M5.5 & 0.024 & $-$7.9$^{a}$ & ... & WTTS & 0.0 & no & ... & $\rm 0.12^{+0.05}_{-0.01}$ & $\rm 4.97^{+6.84}_{-0.75}$ & ... & ... & ... & ...\\
191 &  05 40 46.22 &  $-$08 05 24.3 & K1$^{b}$ & K1 & 13.154 & $-$16.0$^{b}$ & ... & CTTS & 4.8 & yes & $-$1.79 & $\rm 2.34^{+0.31}_{-0.43}$ & $\rm 0.42^{+0.32}_{-0.16}$ & Thick & $-$6.79 & ... & ...\\
192 &  05 40 46.40 &  $-$08 04 36.1 & K7$^{b}$ & K7 & 4.795 & ... & ... & ... & 2.0 & yes & $-$2.48 & $\rm 0.55^{+0.15}_{-0.11}$ & $\rm 0.06^{+4.66}_{-0.03}$ & Debris disk  & ... & ... & ...\\
193 &  05 40 46.62 &  $-$08 07 12.8 & K7$^{b}$ & K7 & 8.229 & ... & ... & ... & 10.0 & yes & $-$2.61 & $\rm 0.57^{+0.13}_{-0.10}$ & $\rm 0.02^{+0.03}_{-0.01}$ & Transition disk & ... & ... & ...\\
194 &  05 40 46.84 &  $-$08 04 54.6 & M2.5$^{b}$ & M2.5 & 1.231 & $-$13.0$^{b}$ & ... & CTTS & 5.2 & yes & 0.08 & $\rm 0.27^{+0.07}_{-0.04}$ & $\rm 0.08^{+0.09}_{-0.04}$ & Thick & ... & ... & ...\\
195 &  05 40 48.07 &  $-$08 05 58.7 & K7$^{b}$ & K7 & 5.832 & ... & ... & ... & 1.9 & yes & $-$1.26 & $\rm 0.55^{+0.13}_{-0.11}$ & $\rm 0.04^{+0.05}_{-0.02}$ & Thick & ... & ... & ...\\
196 &  05 40 57.49 &  $-$07 48 08.8 & M2.5$^{a}$ & M2.5 & 0.566 & $-$37.3$^{a}$ & ... & CTTS & 4.6 & no & $-$1.04 & $\rm 0.32^{+0.20}_{-0.09}$ & $\rm 0.38^{+6.95}_{-0.24}$ & Thick & $-$8.26 & ... & ...\\
197 &  05 40 59.75 &  $-$08 06 03.2 & M6$^{b}$ & M6 & 0.097 & $-$42.0$^{b}$ & ... & CTTS & 0.0 & no & $-$0.35 & $\rm 0.12^{+0.06}_{-0.01}$ & $\rm 0.55^{+1.13}_{-0.11}$ & Thick & $-$9.83 & ... & ...\\
198 &  05 40 59.92 &  $-$07 48 16.1 & M2.5$^{a}$ & M2.5 & 0.187 & $-$65.0$^{a}$ & ... & CTTS & 4.2 & no & $-$1.41 & $\rm 0.40^{+0.28}_{-0.17}$ & $\rm 3.10^{+24.67}_{-2.28}$ & Thick & $-$8.89 & ... & $-$8.94\\
199 &  05 41 03.55 &  $-$07 57 46.3 & M6$^{a}$ & M6 & 0.027 & $-$17.5$^{a}$ & ... & WTTS & 0.0 & no & $-$2.77 & ... & ... & No disk & ... & ... & ...\\
200 &  05 41 04.09 &  $-$07 43 38.6 & M3$^{a}$ & M3 & 0.138 & $-$7.0$^{a}$ & 0.5$^{a}$ & WTTS & 1.5 & no & $-$1.31 & $\rm 0.32^{+0.07}_{-0.06}$ & $\rm 2.99^{+5.19}_{-1.92}$ & Thick & ... & ... & ...\\
201 &  05 41 04.36 &  $-$07 46 40.6 & M5$^{a}$ & M5 & 0.045 & $-$5.7$^{a}$ & ... & WTTS & 0.0 & no & ... & $\rm 0.16^{+0.06}_{-0.03}$ & $\rm 3.93^{+4.75}_{-1.40}$ & ... & ... & ... & ...\\
202 &  05 41 05.03 &  $-$07 45 34.2 & M4.5$^{a}$ & M4.5 & 0.179 & $-$7.7$^{a}$ & ... & WTTS & 2.4 & no & $-$2.67 & $\rm 0.20^{+0.09}_{-0.05}$ & $\rm 0.71^{+1.09}_{-0.34}$ & No disk & ... & ... & ...\\
203 &  05 41 05.49 &  $-$07 47 07.5 & G1$^{a}$ & G1 & 8.105 & $-$53.8$^{a}$ & ... & CTTS & 10.6 & no & $-$1.59 & $\rm 1.95^{+0.16}_{-0.16}$ & $\rm 4.82^{+2.01}_{-1.09}$ & Thick & $-$6.56 & $-$6.51 & ...\\
204 &  05 41 07.00 &  $-$07 47 15.9 & M3.5$^{a}$ & M3.5 & 0.108 & $-$136.6$^{a}$ & ... & CTTS & 0.0 & no & $-$2.07 & $\rm 0.30^{+0.09}_{-0.06}$ & $\rm 3.78^{+2.87}_{-1.73}$ & Transition disk & $-$8.97 & $-$9.20 & ...\\
205 &  05 41 07.08 &  $-$07 46 22.5 & M3.5$^{a}$ & M3.5 & 0.155 & $-$18.0$^{a}$ & ... & CTTS & 1.3 & no & $-$1.32 & $\rm 0.30^{+0.12}_{-0.07}$ & $\rm 2.23^{+6.13}_{-1.16}$ & Thick & $-$9.79 & $-$9.83 & ...\\
206 &  05 41 14.03 &  $-$08 07 57.4 & M2$^{b}$ & M2 & 0.953 & $-$10.0$^{b}$ & ... & WTTS & 1.2 & no & $-$1.25 & $\rm 0.32^{+0.09}_{-0.06}$ & $\rm 0.17^{+1.03}_{-0.09}$ & Thick & ... & ... & ...\\
207 &  05 41 16.25 &  $-$07 43 51.4 & M5.5$^{a}$ & M5.5 & 0.024 & $-$8.0$^{a}$ & ... & WTTS & 0.9 & no & $-$2.69 & $\rm 0.13^{+0.04}_{-0.02}$ & $\rm 6.51^{+6.82}_{-1.92}$ & No disk & ... & ... & ...\\
208 &  05 41 20.12 &  $-$07 55 24.0 & M4$^{a}$ & M4 & 0.581 & $-$48.9$^{a}$ & ... & CTTS & 10.2 & yes & $-$1.41 & $\rm 0.23^{+0.10}_{-0.04}$ & $\rm 0.15^{+0.28}_{-0.08}$ & Thick & $-$8.23 & ... & ...\\
209 &  05 41 25.35 &  $-$08 05 54.7 & K1$^{b}$ & K1 & 4.941 & $-$110.0$^{b}$ & ... & CTTS & 2.6 & no & 0.18 & $\rm 1.94^{+0.10}_{-0.29}$ & $\rm 1.36^{+0.76}_{-0.49}$ & Thick & $-$6.40 & ... & ...\\
210 &  05 41 26.37 &  $-$07 58 17.8 & M5$^{a}$, M6$^{b}$ & M5 & 0.030 & $-$7.0$^{a}$ & 0.6$^{a}$ & WTTS & 0.0 & no & ... & $\rm 0.16^{+0.07}_{-0.03}$ & $\rm 7.09^{+9.96}_{-2.77}$ & ... & ... & ... & ...\\
211 &  05 41 30.59 &  $-$08 04 48.2 & K7$^{b}$ & K7 & 3.357 & $-$40.0$^{b}$ & ... & CTTS & 10.0 & no & $-$0.80 & $\rm 0.55^{+0.13}_{-0.12}$ & $\rm 0.11^{+0.12}_{-0.05}$ & Thick & $-$6.74 & ... & ...\\
212 &  05 41 33.38 &  $-$07 59 56.3 & M1.5$^{a}$ & M1.5 & 1.097 & $-$69.8$^{a}$ & ... & CTTS & 9.7 & no & $-$1.38 & $\rm 0.34^{+0.12}_{-0.07}$ & $\rm 0.17^{+0.23}_{-0.08}$ & Thick & $-$7.31 & ... & ...\\
213 &  05 41 41.66 &  $-$08 00 18.5 & K7.5$^{a}$ & K7.5 & 0.047 & $-$28.4$^{a}$ & ... & CTTS & 4.5 & no & $-$1.02 & ... & ... & Thick & ... & ... & ...\\
214 &  05 41 43.74 &  $-$07 58 22.3 & M5.5$^{a}$ & M5.5 & 0.468 & $-$405.3$^{a}$ & 0.4$^{a}$ & CTTS & 2.4 & no & $-$1.01 & $\rm 0.16^{+0.03}_{-0.01}$ & $\rm 0.04^{+0.09}_{-0.02}$ & Thick & $-$7.49 & $-$8.40 & $-$8.68\\
215 &  05 41 49.23 &  $-$07 57 12.3 & M3.5$^{a}$ & M3.5 & 0.150 & $-$18.5$^{a}$ & ... & CTTS & 3.1 & no & $-$1.53 & $\rm 0.28^{+0.09}_{-0.06}$ & $\rm 2.09^{+4.47}_{-0.91}$ & Thick & $-$9.80 & ... & ...\\
216 &  05 41 49.74 &  $-$08 00 32.3 & K5$^{b}$ & K5 & 8.432 & $-$18.0$^{b}$ & ... & CTTS & 0.4 & no & $-$0.94 & $\rm 0.86^{+0.36}_{-0.16}$ & $\rm 0.07^{+0.14}_{-0.03}$ & Thick & $-$6.69 & ... & ...\\
217 &  05 41 51.84 &  $-$08 06 56.6 & M4.5$^{a}$ & M4.5 & 0.129 & $-$5.5$^{a}$ & 0.7$^{a}$ & WTTS & 2.7 & no & $-$2.70 & $\rm 0.20^{+0.02}_{-0.02}$ & $\rm 1.23^{+0.25}_{-0.22}$ & No disk & ... & ... & ...\\
218 &  05 41 54.66 &  $-$07 59 12.4 & K3.5$^{a}$, K8$^{b}$ & K3.5 & 2.039 & $-$6.6$^{a}$ & 0.6$^{a}$ & CTTS & 5.9 & no & $-$1.73 & $\rm 1.20^{+0.24}_{-0.37}$ & $\rm 1.42^{+1.63}_{-0.75}$ & Thick & $-$8.40 & ... & ...\\
219 &  05 41 55.96 &  $-$07 58 36.6 & M4$^{a}$ & M4 & 0.063 & $-$4.1$^{a}$ & ... & WTTS & 3.8 & no & $-$2.04 & $\rm 0.23^{+0.12}_{-0.06}$ & $\rm 4.83^{+9.01}_{-2.24}$ & Thin & ... & ... & ...\\
220 &  05 42 05.59 &  $-$08 01 05.6 & M2$^{b}$ & M2 & 0.192 & ... & ... & ... & 0.9 & no & $-$1.76 & $\rm 0.43^{+0.13}_{-0.10}$ & $\rm 3.51^{+11.00}_{-1.73}$ & Transition disk & ...&...&...\\[2pt]
\hline
221 &  05 42 07.25 &  $-$08 05 24.4 & M6$^{b}$ & M6 & 0.049 & ... & ... & ... & 1.6 & no & $-$1.18 & $\rm 0.10^{+0.06}_{-0.00}$ & $\rm 1.27^{+3.53}_{-0.46}$ & Thick & ... & ... & ...\\
222 &  05 42 10.51 &  $-$08 07 35.1 & M5$^{a}$, M4$^{b}$ & M5 & 0.084 & $-$7.8$^{a}$ & ... & WTTS & 0.0 & no & ... & $\rm 0.18^{+0.04}_{-0.03}$ & $\rm 1.84^{+1.07}_{-0.62}$ & ... & ... & ... & ...\\
223 &  05 42 11.04 &  $-$08 01 05.4 & M4.5$^{a}$, M3$^{b}$ & M4.5 & 0.056 & ... & 0.5$^{a}$ & ... & 0.0 & no & $-$1.16 & $\rm 0.19^{+0.05}_{-0.03}$ & $\rm 3.87^{+3.06}_{-1.34}$ & Thick & ... & ... & $-$10.98\\
224 &  05 42 13.51 &  $-$08 10 01.5 & M3$^{b}$ & M3 & 0.080 & $-$50.0$^{b}$ & ... & CTTS & 1.6 & no & $-$1.09 & $\rm 0.35^{+0.12}_{-0.10}$ & $\rm 8.43^{+22.49}_{-4.73}$ & Thick & $-$9.76 & ... & ...\\
225 &  05 42 14.60 &  $-$07 58 57.9 & M3.5$^{a}$ & M3.5 & 0.200 & $-$13.2$^{a}$ & 0.6$^{a}$ & WTTS & 1.1 & no & $-$1.09 & $\rm 0.29^{+0.05}_{-0.04}$ & $\rm 1.36^{+1.08}_{-0.42}$ & Thick & ... & ... & ...\\
226 &  05 42 15.93 &  $-$08 09 37.8 & M5$^{a}$ & M5 & 0.134 & $-$6.3$^{a}$ & ... & WTTS & 3.3 & no & $-$2.31 & $\rm 0.18^{+0.06}_{-0.04}$ & $\rm 0.85^{+0.96}_{-0.36}$ & Thin & ... & ... & ...\\
227 &  05 42 18.11 &  $-$08 08 10.9 & K4.5$^{a}$, K5$^{b}$ & K4.5 & 0.924 & $-$1.0$^{a}$ & 0.5$^{a}$ & WTTS & 1.5 & yes & $-$2.96 & $\rm 1.13^{+0.09}_{-0.28}$ & $\rm 3.89^{+4.95}_{-1.95}$ & No disk & ... & ... & ...\\
228 &  05 42 18.56 &  $-$07 59 23.0 & M4.5$^{a}$ & M4.5 & 0.069 & $-$6.9$^{a}$ & 0.6$^{a}$ & WTTS & 0.0 & no & $-$2.74 & $\rm 0.19^{+0.02}_{-0.01}$ & $\rm 2.68^{+0.56}_{-0.43}$ & No disk & ... & ... & ...\\
229 &  05 42 19.05 &  $-$07 59 12.2 & M5.5$^{a}$ & M5.5 & 0.041 & $-$18.9$^{a}$ & 0.8$^{a}$ & CTTS & 0.0 & no & ... & $\rm 0.11^{+0.04}_{-0.01}$ & $\rm 2.09^{+2.29}_{-0.46}$ & ... & $-$10.87 & ... & ...\\
230 &  05 42 19.56 &  $-$08 05 07.6 & M2.5$^{b}$ & M2.5 & 0.134 & ... & ... & ... & 2.0 & no & $-$1.31 & $\rm 0.39^{+0.13}_{-0.10}$ & $\rm 4.88^{+13.33}_{-2.62}$ & Thick & ... & ... & ...\\
231 &  05 42 21.36 &  $-$08 06 47.9 & M3.5$^{a}$ & M3.5 & 0.237 & $-$12.9$^{a}$ & ... & WTTS & 4.8 & no & $-$2.57 & $\rm 0.26^{+0.04}_{-0.03}$ & $\rm 0.88^{+0.41}_{-0.24}$ & No disk & ... & ... & ...\\
232 &  05 42 21.56 &  $-$08 08 29.8 & M0$^{a}$, K8$^{b}$ & M0 & 0.464 & $-$39.3$^{a}$, $-$10.0$^{b}$ & 0.5$^{a}$ & CTTS & 0.5 & yes & $-$2.12 & $\rm 0.58^{+0.12}_{-0.08}$ & $\rm 1.95^{+2.21}_{-0.89}$ & Transition disk & $-$8.35 & $-$8.42 & $-$7.98\\
233 &  05 42 23.24 &  $-$07 58 28.9 & M5$^{a}$ & M5 & 0.034 & $-$5.9$^{a}$ & 0.7$^{a}$ & WTTS & 0.1 & no & ... & $\rm 0.17^{+0.05}_{-0.03}$ & $\rm 6.87^{+5.77}_{-2.54}$ & ... & ... & ... & ...\\
234 &  05 42 23.61 &  $-$08 09 10.3 & M3$^{a}$, M2$^{b}$ & M3 & 0.248 & $-$44.5$^{a}$, $-$65.0$^{b}$ & 0.9$^{a}$ & CTTS & 1.1 & yes & $-$1.49 & $\rm 0.30^{+0.05}_{-0.04}$ & $\rm 1.14^{+0.63}_{-0.32}$ & Thick & $-$8.91 & $-$9.07 & $-$8.75\\
235 &  05 42 23.94 &  $-$08 09 45.9 & K4$^{a}$, K5$^{b}$ & K4 & 1.197 & $-$0.9$^{a}$ & 0.4$^{a}$ & WTTS & 3.1 & yes & $-$2.88 & $\rm 1.23^{+0.10}_{-0.32}$ & $\rm 3.42^{+4.88}_{-1.81}$ & No disk & ... & ... & ...\\
236 &  05 42 25.32 &  $-$08 10 18.0 & M1$^{a}$, M0$^{b}$ & M1 & 0.420 & $-$3.3$^{a}$ & 0.7$^{a}$ & WTTS & 0.7 & yes & $-$2.82 & $\rm 0.48^{+0.33}_{-0.18}$ & $\rm 1.47^{+9.16}_{-0.96}$ & No disk & ... & ... & ...\\
237 &  05 42 25.44 &  $-$08 09 55.4 & M1$^{a}$, K8$^{b}$ & M1 & 0.216 & $-$21.2$^{a}$, $-$27.0$^{b}$ & ... & CTTS & 1.8 & yes & $-$1.66 & $\rm 0.55^{+0.11}_{-0.09}$ & $\rm 5.10^{+7.10}_{-2.07}$ & Thick & $-$9.34 & $-$9.38 & $-$9.40\\
238 &  05 42 25.68 &  $-$07 58 22.0 & M3$^{a}$ & M3 & 0.135 & $-$3.3$^{a}$ & 0.2$^{a}$ & WTTS & 0.5 & no & $-$2.60 & $\rm 0.34^{+0.05}_{-0.04}$ & $\rm 3.66^{+1.69}_{-0.99}$ & No disk & ... & ... & ...\\
239 &  05 42 25.79 &  $-$08 08 50.1 & M2.5$^{a}$ & M2.5 & 0.210 & $-$61.8$^{a}$ & 0.5$^{a}$ & CTTS & 1.2 & yes & $-$1.30 & $\rm 0.34^{+0.04}_{-0.04}$ & $\rm 1.89^{+0.77}_{-0.44}$ & Thick & $-$8.87 & $-$9.33 & $-$9.32\\
240 &  05 42 26.50 &  $-$07 58 50.9 & K6$^{a}$ & K6 & 1.542 & $-$91.6$^{a}$ & 0.4$^{a}$ & CTTS & 4.4 & no & $-$0.64 & $\rm 0.69^{+0.49}_{-0.30}$ & $\rm 0.57^{+11.54}_{-0.40}$ & Thick & $-$6.95 & ... & ...\\
241 &  05 42 26.96 &  $-$08 09 17.3 & M3.5$^{a}$ & M3.5 & 0.109 & $-$3.4$^{a}$ & ... & WTTS & 0.0 & yes & ... & $\rm 0.31^{+0.03}_{-0.02}$ & $\rm 4.08^{+0.91}_{-0.64}$ & ... & ... & ... & ...\\
242 &  05 42 27.50 &  $-$08 11 03.8 & M5$^{a}$, M3$^{b}$ & M5 & 0.034 & $-$9.4$^{a}$ & ... & WTTS & 0.0 & yes & $-$2.09 & $\rm 0.15^{+0.04}_{-0.02}$ & $\rm 4.92^{+3.31}_{-1.71}$ & Transition disk & ... & ... & ...\\
243 &  05 42 27.59 &  $-$08 09 52.6 & M1$^{b}$ & M1 & 0.294 & $-$13.0$^{b}$ & ... & CTTS & 1.1 & yes & $-$1.20 & $\rm 0.51^{+0.15}_{-0.11}$ & $\rm 2.82^{+9.09}_{-1.30}$ & Thick & $-$9.33 & ... & ...\\
244 &  05 42 29.78 &  $-$08 07 26.4 & M1$^{a}$ & M1 & 0.372 & $-$3.4$^{a}$ & 0.6$^{a}$ & WTTS & 2.8 & yes & ... & $\rm 0.50^{+0.44}_{-0.23}$ & $\rm 1.92^{+22.80}_{-1.49}$ & ... & ... & ... & ...\\
245 &  05 42 30.62 &  $-$07 59 54.0 & M2$^{a}$ & M2 & 0.108 & $-$2.2$^{a}$ & 0.5$^{a}$ & WTTS & 0.1 & no & $-$2.87 & $\rm 0.47^{+0.10}_{-0.11}$ & $\rm 10.32^{+20.81}_{-4.36}$ & No disk & ... & ... & ...\\
246 &  05 42 30.87 &  $-$08 10 45.7 & M5$^{a}$ & M5 & 0.090 & $-$5.0$^{a}$ & 0.6$^{a}$ & WTTS & 0.9 & yes & $-$1.64 & $\rm 0.18^{+0.03}_{-0.02}$ & $\rm 1.71^{+0.65}_{-0.44}$ & Transition disk & ... & ... & ...\\
247 &  05 42 31.00 &  $-$08 06 38.1 & M3.5$^{b}$ & M3.5 & 0.166 & $-$30.0$^{b}$ & ... & CTTS & 0.0 & yes & $-$1.72 & $\rm 0.28^{+0.10}_{-0.07}$ & $\rm 1.71^{+2.01}_{-0.91}$ & Thick & $-$9.46 & ... & ...\\
248 &  05 42 31.33 &  $-$08 02 35.3 & M4.5$^{a}$ & M4.5 & 0.086 & $-$8.3$^{a}$ & 0.7$^{a}$ & WTTS & 0.0 & no & $-$2.79 & $\rm 0.19^{+0.03}_{-0.02}$ & $\rm 2.12^{+0.72}_{-0.50}$ & No disk & ... & ... & ...\\
249 &  05 42 31.55 &  $-$08 08 22.8 & M4$^{a}$, M2$^{b}$ & M4 & 0.270 & $-$12.4$^{a}$, $-$15.0$^{b}$ & ... & WTTS & 2.8 & yes & $-$0.60 & $\rm 0.22^{+0.04}_{-0.04}$ & $\rm 0.47^{+0.38}_{-0.20}$ & Thick & ... & ... & ...\\
250 &  05 42 32.66 &  $-$08 10 07.9 & M4.5$^{a}$ & M4.5 & 0.119 & $-$6.7$^{a}$ & ... & WTTS & 1.5 & yes & ... & $\rm 0.20^{+0.02}_{-0.02}$ & $\rm 1.41^{+0.48}_{-0.32}$ & ... & ... & ... & ...\\
251 &  05 42 32.77 &  $-$08 07 04.8 & M5$^{a}$ & M5 & 0.132 & $-$11.0$^{a}$ & ... & WTTS & 0.3 & yes & ... & $\rm 0.19^{+0.02}_{-0.01}$ & $\rm 1.02^{+0.32}_{-0.20}$ & ... & ... & ... & ...\\
252 &  05 42 34.02 &  $-$08 09 59.7 & K4$^{a}$, K7$^{b}$ & K4 & 2.186 & $-$1.9$^{a}$ & 0.5$^{a}$ & WTTS & 5.0 & yes & $-$2.33 & $\rm 1.17^{+0.26}_{-0.38}$ & $\rm 1.22^{+1.36}_{-0.67}$ & Transition disk & ... & ... & ...\\
253 &  05 42 34.31 &  $-$08 02 12.1 & M5$^{a}$, M4$^{b}$ & M5 & 0.068 & $-$86.0$^{a}$, $-$48.0$^{b}$ & 0.4$^{a}$ & CTTS & 0.0 & no & $-$1.28 & $\rm 0.16^{+0.24}_{-0.04}$ & $\rm 1.99^{+29.05}_{-0.78}$ & Thick & $-$9.68 & ... & $-$9.98\\
254 &  05 42 34.92 &  $-$08 07 50.5 & M5$^{a}$ & M5 & 0.126 & $-$6.0$^{a}$ & ... & WTTS & 1.9 & yes & ... & $\rm 0.19^{+0.05}_{-0.04}$ & $\rm 1.08^{+1.19}_{-0.43}$ & ... & ... & ... & ...\\
255 &  05 42 36.41 &  $-$08 10 17.8 & M3$^{b}$ & M3 & 0.063 & $-$36.0$^{b}$ & ... & CTTS & 2.8 & yes & ... & $\rm 0.36^{+0.12}_{-0.10}$ & $\rm 12.49^{+16.77}_{-6.16}$ & ... & $-$10.12 & ... & ...\\
256 &  05 42 36.51 &  $-$08 11 51.5 & K5$^{a}$ & K5 & 0.687 & $-$1.7$^{a}$ & 0.5$^{a}$ & WTTS & 3.0 & no & ... & $\rm 1.06^{+0.06}_{-0.20}$ & $\rm 5.06^{+7.10}_{-2.17}$ & ... & ... & ... & ...\\[2pt]
\hline
257& 05 42 40.06& $-$08 05 07.7&M2$^{b}$&M2&0.086&$-$12.0$^{b}$ & ... & CTTS & 0.4 & no & $-$1.85 & $\rm 0.49^{+0.09}_{-0.13}$ & $\rm 15.81^{+16.78}_{-7.46}$ & Transition disk & $-$10.43 & ... & ...\\
258 &  05 42 42.10 &  $-$08 15 15.3 & M1$^{b}$ & M1 & 0.356 & ... & ... & ... & 3.5 & no & $-$1.18 & $\rm 0.49^{+0.14}_{-0.11}$ & $\rm 1.95^{+4.72}_{-0.75}$ & Thick & ... & ... & ...\\
259 &  05 42 44.12 &  $-$08 06 26.4 & M3$^{b}$ & M3 & 0.251 & $-$22.0$^{b}$ & ... & CTTS & 0.2 & no & $-$1.22 & $\rm 0.31^{+0.10}_{-0.07}$ & $\rm 1.15^{+1.16}_{-0.49}$ & Thick & $-$9.28 & ... & ...\\
260 &  05 42 47.18 &  $-$08 05 57.0 & M2$^{b}$ & M2 & 0.110 & ... & ... & ... & 0.2 & no & $-$1.99 & $\rm 0.47^{+0.12}_{-0.13}$ & $\rm 9.97^{+9.87}_{-4.64}$ & Transition disk & ... & ... & ...\\
261 &  05 42 47.44 &  $-$08 04 05.4 & M2.5$^{b}$ & M2.5 & 0.115 & $-$76.0$^{b}$ & ... & CTTS & 0.8 & no & $-$0.87 & $\rm 0.40^{+0.13}_{-0.10}$ & $\rm 6.39^{+13.83}_{-3.36}$ & Thick & $-$9.23 & ... & ...\\
262 &  05 42 48.24 &  $-$08 05 57.1 & M1$^{b}$ & M1 & 0.088 & $-$24.0$^{b}$ & ... & CTTS & 0.8 & no & ... & $\rm 0.61^{+0.03}_{-0.12}$ & $\rm 28.01^{+22.19}_{-12.32}$ & ... & $-$9.99 & ... & ...\\
263 &  05 42 51.65 &  $-$08 07 41.2 & M1$^{b}$ & M1 & 0.093 & $-$34.0$^{b}$ & ... & CTTS & 2.0 & no & $-$1.56 & $\rm 0.61^{+0.03}_{-0.12}$ & $\rm 25.51^{+20.61}_{-10.84}$ & Thick & $-$9.76 & ... & ...\\
264 &  05 42 52.30 &  $-$08 08 54.9 & M5$^{b}$ & M5 & 0.065 & ... & ... & ... & 0.6 & no & $-$1.38 & $\rm 0.17^{+0.09}_{-0.04}$ & $\rm 2.37^{+5.70}_{-0.94}$ & Thick & ... & ... & ...\\
265 &  05 42 59.13 &  $-$08 09 23.6 & M1$^{b}$ & M1 & 0.389 & $-$67.0$^{b}$ & ... & CTTS & 2.0 & no & $-$0.88 & $\rm 0.48^{+0.14}_{-0.10}$ & $\rm 1.63^{+5.27}_{-0.62}$ & Thick & $-$8.20 & ... & ...\\
266 &  05 43 01.78 &  $-$08 04 50.6 & K5$^{b}$ & K5 & 1.869 & $-$32.0$^{b}$ & ... & CTTS & 2.7 & no & $-$2.12 & $\rm 0.86^{+0.29}_{-0.15}$ & $\rm 0.73^{+0.99}_{-0.24}$ & Transition disk & $-$7.52 & ... & ...\\
267 &  05 43 08.49 &  $-$08 11 32.3 & M2$^{b}$ & M2 & 0.053 & ... & ... & ... & 0.1 & no & $-$1.59 & $\rm 0.50^{+0.06}_{-0.13}$ & $\rm 36.34^{+24.57}_{-19.52}$ & Thick & ... & ... & ...\\[2pt]
\hline
\end{tabular}
\end{table}